

TOPICAL REVIEW • OPEN ACCESS

Deep learning-based spike sorting: a survey

To cite this article: Luca M Meyer et al 2024 J. Neural Eng. 21 061003

View the article online for updates and enhancements.

You may also like

- <u>Clustering for mitigating subject variability</u> in driving fatigue classification using electroencephalography source-space functional connectivity features Khanh Ha Nguyen, Yvonne Tran, Ashley Craig et al.
- <u>NeuSort: an automatic adaptive spike</u> sorting approach with neuromorphic models Hang Yu, Yu Qi and Gang Pan
- Towards online spike sorting for highdensity neural probes using discriminative template matching with suppression of interfering spikes Jasper Wouters, Fabian Kloosterman and Alexander Bertrand



This content was downloaded from IP address 152.78.0.24 on 26/11/2024 at 16:24

Journal of Neural Engineering



OPEN ACCESS

RECEIVED

8 March 2024

REVISED 1 October 2024

ACCEPTED FOR PUBLICATION 25 October 2024

PUBLISHED 14 November 2024

Original content from this work may be used under the terms of the Creative Commons Attribution 4.0 licence.

Any further distribution of this work must maintain attribution to the author(s) and the title of the work, journal citation and DOI.



TOPICAL REVIEW

Deep learning-based spike sorting: a survey

Luca M Meyer¹, Majid Zamani², János Rokai³, and Andreas Demosthenous^{4,*}

- ¹ Currently not Affiliated with any Institution, Wiesbaden, Germany
- ² School of Electronics and Computer Science, University of Southampton, Southampton, United Kingdom
- ³ Institute of Cognitive Neurosciences and Psychology, Research Centre for Natural Sciences, Budapest, Hungary
- ⁴ Department of Electronic and Electrical Engineering, University College London, London, United Kingdom
 - Author to whom any correspondence should be addressed.

E-mail: a.demosthenous@ucl.ac.uk, lucamaximilianmeyer@icloud.com, m.zamani@soton.ac.uk and rokai.janos@ttk.hu

Keywords: deep learning, feature extraction, neural networks, spike detection, spike classification, spike sorting

Abstract

Objective. Deep learning is increasingly permeating neuroscience, leading to a rise in signal-processing applications for extracellular recordings. These signals capture the activity of small neuronal populations, necessitating 'spike sorting' to assign action potentials (spikes) to their underlying neurons. With the rise in publications delving into new methodologies and techniques for deep learning-based spike sorting, it is crucial to synthesise these findings critically. This survey provides an in-depth evaluation of the approaches, methodologies and outcomes presented in recent articles, shedding light on the current state-of-the-art. Approach. Twenty-four articles published until December 2023 on deep learning-based spike sorting have been examined. The proposed methods are divided into three sub-problems of spike sorting: spike detection, feature extraction and classification. Moreover, integrated systems, i.e. models that detect spikes and extract features or do classification within a single network, are included. Main results. Although most algorithms have been developed for single-channel recordings, models utilising multi-channel data have already shown promising results, with efficient hardware implementations running quantised models on application-specific integrated circuits and field programmable gate arrays. Convolutional neural networks have been used extensively for spike detection and classification as the data can be processed spatiotemporally while maintaining low-parameter models and increasing generalisation and efficiency. Autoencoders have been mainly utilised for dimensionality reduction, enabling subsequent clustering with standard methods. Also, integrated systems have shown great potential in solving the spike sorting problem from end to end. Significance. This survey explores recent articles on deep learning-based spike sorting and highlights the capabilities of deep neural networks in overcoming associated challenges, but also highlights potential biases of certain models. Serving as a resource for both newcomers and seasoned researchers in the field, this work provides insights into the latest advancements and may inspire future model development.

1. Introduction

Neurons communicate through the exchange of electrical impulses, more precisely, through the release of neurotransmitters, which results in the propagation of action potentials [1]. Signals of individual neurons and the signal exchange with their neighbours are usually investigated with extracellular recordings. Extracellular electrodes typically capture the activity of a small group of neurons. According to an approximation by Pedreira *et al* [2], which is based on the assumption that there are 300 000 neurons mm³ (as in the rat hippocampus [3]), following Coulomb's law for amplitude decay as $v \sim 1/r^2$, and assuming that action potentials of ~60 μ V are generated at a distance of 50 μ m [4] (maximum distance and minimum amplitude to identify single spikes), more than thirty neurons would generate spikes with an amplitude ranging from 60 μ V to 70 μ V, and around ten neurons would generate amplitudes between 100 μ V and 110 μ V. However, the measured signals are contaminated by noise originating from neurons not

close to the electrode. Additionally, artefacts originating from the recording equipment influence the recorded signal, resulting in fewer spikes reaching the above voltage levels. Neuronal waveforms exhibit distinct characteristics, primarily determined by the cell type, ion channel distribution, dendritic tree morphology, and the orientation and distance to the recording electrode [5, 6]. These features enable spike sorting, i.e. extracting spikes from neural recordings and assigning them to their underlying neurons. Multi-electrode arrays (MEA), such as tetrodes or polytrodes, enable the utilisation of spatial spike features, e.g. by triangulation [4]. High-density MEA (HD-MEA) recordings with hundreds of channels are already in use [7] and with the growing number of neurons being measured, it has become imperative to employ automatic, high-bandwidth spike sorting algorithms to analyse single neuron activity at scale. Advancements in spike sorting and HD-MEA technologies are essential for neuroscience research, clinical applications, and brain-computer interfaces (BCIs). In fundamental neuroscience, spike sorting offers valuable insights into the functioning of neuronal circuits, advancing our understanding of the interactions between cerebral regions and behaviours. Clinically, the analysis of extracellular recordings is vital for studying neurological conditions, including epilepsy [8], paralysis [8, 9], and cognitive loss [10]. Spike sorting is also crucial for neural signal decoding and their translation into commands for external device control in BCIs and neuroprosthetics [5]. It has been shown that single-neuron activity can help to decode movement [8], intentions [11], and memory [10].

Deep neural networks (DNNs) are becoming increasingly important in neuroscience. In neuroimaging, deep learning is used to detect and analyse Alzheimer's disease [12], brain strokes [13], autism [14], and other neurological and psychiatric disorders [15]. Deep learning models enhance neuroimaging techniques by automating the segmentation of brain images, for example, to detect brain tumors [16] or improve signal extraction from noisy data [17]. Also, deep learning is used to model neurodegenerative diseases and analyse their underlying mechanisms [18]. Other deep learning applications in neuroscience research include decoding brain activity on a macro level, e.g. by analysing functional magnetic resonance images [19] or electroencephalogram (EEG) [20], and on a micro level, by analysing intracranial micro-electrode recordings, e.g. for spike sorting [21]. For the latter, deep learning offers the potential to handle high-dimensional data, perform automatic feature learning, and improve system scalability and noise robustness. For decades, attempts have been made to solve the spike sorting problem, and many reviews have tried to provide insights on the stateof-the-art [4, 22-25]. Recent reviews on spike sorting

methods [26–29] do not consider DNN-based models in detail. To address this gap, this survey provides a comprehensive evaluation of current approaches on deep learning-based spike sorting.

The first section of this paper provides a brief overview of conventional spike sorting methods and different types of data commonly used for their evaluation. Also, the challenges of spike sorting and requirements for modern spike sorting algorithms are outlined. Section 2 focuses on state-of-the-art spike sorting methods that are not based on neural networks (NNs), followed by brief historical developments related to the application of NNs in spike sorting. Subsequently, two innovative studies are examined to demonstrate that NNs with a single hidden layer can be used in various spike sorting scenarios, showcasing both the limits and advantages of such shallow models. The criteria for the selected publications in this survey are also outlined. Section 3 focuses on spike detection, section 4 deals with feature extraction, section 5 covers classification models, and section 6 focuses on integrated systems, i.e. DNNs that detect spikes and perform feature extraction or classification using a single model. Section 7 critically discusses how state-of-the-art deep learningbased spike sorting models address the challenges and requirements described in section 1.3, identifies future directions and outlines hardware implementations of certain models. Concluding remarks are drawn in section 8.

1.1. Conventional spike sorting

Traditionally, the spike sorting procedure is separated into several steps, as shown in figure 1. Using a band-pass filter on the raw data, the signal of interest consisting of neural activities can be isolated. Single spikes can be detected with techniques like adaptive amplitude or power thresholding. As these methods may lead to inclusions of non-neural activity (false-positives) and spikes still can be missed (false-negatives), other techniques, such as the non-linear energy operator [30] or the continuous wavelet transformation [31], have been applied extensively to detect spikes. Once detected, spikes are typically aligned to their amplitude peaks for subsequent analysis [24], and features are extracted in order to reduce dimensions. Standard feature extraction methods include principal component analysis (PCA) [32], independent component analysis [33], t-distributed stochastic neighbour embedding (t-SNE) [34], and discrete wavelet transform [35]. Subsequently, the extracted spike features are clustered, often using the k-means algorithm [36], where each centroid represents a template for a particular neuron. While k-means requires k as a parameter for the number of clusters in the data, density-based spatial clustering of applications with noise (DBSCAN) [37] does not require information



sorting. (a) Four electrodes capture changes in the electrical activity of three neurons near to the electrode. The raw data (b) typically are processed with a band-pass filter (c), which enables spike detection (d). Subsequently, the spikes are aligned (e) to extract features (f). Clustering can be applied to the low level feature space to group the spiking events (g), where the average of one cluster is called template (h). Finally, the spike trains can be reconstructed with the corresponding spike times (i). With permission from [23]. Reprinted from [23], Copyright (2011), with permission from Elsevier.

about the number of clusters but often fails to identify clusters of varying density and requires careful hyperparameter tuning. Hierarchical [38] and Bayesian clustering [39] are also frequently used to cluster neural data. Spike trains, i.e. binary signals that indicate the temporal firings of a single neuron, are the desired outcome of spike sorting and are generated with the obtained spike times for each cluster. The explained pipeline represents unsupervised spike sorting, where data patterns and similarities within the data are identified to group similar spike waveforms. However, supervised learning is increasingly being used in machine learning-based spike sorting, where support vector machines [40] and NNs [41, 42] trained with labelled data can be used for both spike detection and classification, effectively mapping spikes to their underlying neurons.

1.2. Datasets

Three types of datasets are commonly used to evaluate spike sorting algorithms: (i) real extracellular recordings, (ii) synthetic datasets, and iii) hybrid datasets. Among these datasets, real extracellular recordings are the most challenging, as they encapsulate a myriad of real-world challenges, such as artefacts or multi-component noise, rendering them intricate but invaluable for advanced analyses. A significant limitation of real data is the absence of ground truth (GT). Patch clamp recordings [43] are frequently employed in this particular setting due to their capability to offer a precise and direct assessment of neuronal activity, thus serving as a dependable reference point. However, they require a significantly greater effort in recording. Thus, synthetic datasets have ascended as a standard for evaluation. Synthetic data are meticulously constructed and typically mimic a local field

potential (LFP), often generated by numerous overlapping spike templates, upon which various spike templates are subsequently superimposed at a larger scale to emulate the activity of a small number of neurons in the vicinity of the electrode [44]. Despite their utility, synthetic datasets have inherent limitations, as they often lack the incorporation of practical scenarios such as bursting neurons, electrode drift, or transient noise. To address these gaps, hybrid datasets have been employed frequently, combining the strengths of real and synthetic data for more accurate analysis and interpretation. Various platforms, such as Collaborative Research in Computational Neuroscience [45], SpikeForest [46], or individual labs with public data repositories [47, 48] provide access to large data from several species and brain regions acquired with various electrode types.

1.3. Challenges and requirements

Transient signals are one of the biggest challenges in spike sorting, as the recorded noise level may fluctuate [49], and spike waveforms often vary in their expression [5, 6]. Since the neuron-electrode distance and orientation can change during a recording, the measured voltages may vary over time, resulting in electrode drift [24]. Moreover, neurons exhibit burst firing patterns, leading to an amplitude decrease of up to 80% [4]. In 2012, Pedreira et al discussed several reasons why the amount of identified neurons per channel appeared lower than expected with existing spike sorters, pointing to sparsely firing neurons as one possible reason [2]. Such neurons pose a significant challenge, especially for template-matching approaches, as they require a specific template for each firing neuron. This, along with transient waveforms, makes adaptive algorithms necessary. Since each channel in an extracellular recording captures the activity of several neurons, two or more neurons may fire simultaneously, leading to overlapping waveforms in the recording. Spike overlaps have been studied for decades and yet have not been fully resolved [50–52]. Moreover, spike sorting models are usually subject to the missing ground-truth problem, as the number of firing neurons during a recording is typically unknown in real-world situations. Furthermore, modern spike sorting models must be scalable to handle HD-MEA recordings. Resource-efficient data processing methods are therefore required for hardware implementation without jeopardising online processing. Implantable on-chip processors for spike sorting are particularly advantageous because they can massively reduce the data rate to be transferred, which directly impacts the power consumption of resource-limited brain chips. On-chip compression models are an alternative approach to reduce the data that needs to be transferred. Apart from that, robust spike sorting algorithms must be transferable for use on different subjects, brain areas, and electrode types.

IOP Publishing

Given the practical limitations, the capacity to generalise across various setups is not just convenient but essential. Reproducibility is another requirement of modern algorithms. Spike sorting models often involve many hyperparameters that require careful tuning for optimal performance. In order to accelerate collaborative progress, it is therefore important that new models (code and data repositories) are published open access.

2. Background and literature overview

2.1. State-of-the-art spike sorting

Superparamagnetic clustering (SPC) was introduced to spike sorting by Quiroga et al in 2004 [44] and is still used as a benchmark. In SPC, data are grouped by evaluating spin-spin correlations derived from the ferromagnetic model of the data. SPC sorts spikes in an unsupervised fashion and provides high accuracy at the expense of computation. Osort [53], one of the first advanced spike sorting algorithms for online processing, is another method that does not require knowledge of the number of clusters in the data. Other online sorting algorithms include GEMsort [54], an on-implant sorter that is based on salient feature selection [55] and Geo-OSort [56] which are mentioned here to increase the technical depth of this survey. Recent advancements in HD-MEA have increased the signal quality, leading to a rise in detectable action potentials with more overlapping spikes [57]. As a result, source separation models have emerged as the preferred approach to process multichannel recordings. Kilosort [21], one such model, processes neural data in several stages, including preprocessing, clustering, template-matching and postprocessing. Kilosort employs a modified k-means clustering technique that is robust to amplitude alterations in spikes. This improves its robustness and flexibility in handling different signal fluctuations, thus enhancing its performance in spike sorting tasks. In 2024, Pachitariu et al proposed Kilosort4, which could improve the clustering stage using a graphbased approach [58]. Kilosort4 is recognised for its ability to accurately detect neurons with low amplitudes and small spatial extents, even under conditions of high drift, which is essential to accurately identify neuronal spikes in challenging recording conditions [58]. Despite the advantages of Kilosort, there are also limitations, which may be affected by various factors, including electrode density, signalto-noise ratio (SNR) and computational resources, potentially providing suboptimal results in specific conditions. Furthermore, the Kilosort algorithm has tuneable parameters. Although good results can be achieved with the default settings, adjusting the parameters can introduce bias. In 2017, MountainSort [59], another advanced spike sorting system for HD-MEA data, was introduced by Chung et al. MountainSort has also been iteratively improved

since its introduction, leading to the recent release of MountainSort4 [60], an efficient spike sorting algorithm that first over-clusters spikes based on the geometric layout of the electrode array and then merges clusters across adjacent channels to remove redundancy. One of its main benefits is its model independence, enabling it to efficiently manage variations in non-Gaussian waveforms under various recording conditions [59]. Compared to Kilosort, this algorithm is fully automatic at the expense of accuracy. Other popular algorithms reported in recent years are KlustaKwik [61], SpyKING CIRCUS [62], and IronClust [63]. Recently, Pachitariu et al reported that IronClust is the nearest competing algorithm to Kilosort4, especially under electrode drift. At the same time, SpyKING CIRCUS and MountainSort4 performed similarly to IronClust on non-drifting recordings [64]. Although Kilosort4 is perhaps one of the best contemporary solutions, there is still scope for improvement; e.g. fast electrode drift still poses a problem [64].

2.2. Spike sorting with NNs

NN-based spike sorting dates back to 1988, when Bower et al used a Hopfield network to associate memory with binary thresholds for spike classification [65]. While the 1990s saw advancements with single-hidden-layer NNs (SHL-NNs) for spike detection and classification [66-68], the field continued to evolve, leading to the exploration of selforganising maps [69-71], known for their unsupervised learning capabilities. The quest for computational efficiency in spike sorting led to the emergence of spiking NNs (SNNs) [72-75], which mirror the time-dependent nature of biological NNs more closely. Running on neuromorphic hardware, SNNs can exploit their full potential, offering benefits in terms of energy efficiency due to sparse, event-driven computations and the ability to recognise spatiotemporal data patterns [76], but often depend on precise hyperparameter tuning [27].

Several SHL-NNs targeting specific tasks in the spike-sorting pipeline have been proposed recently. In the following, two such solutions are briefly described. In 2020, Issar et al [41] proposed a multilayer perceptron (MLP) to separate spikes from noise that impaired decoding. A supervised learning model was developed to predict the probability that a given waveform is a spike [41]. The model was trained with 24 810 795 labelled waveforms derived from extracellular recordings from four monkeys, two brain regions, two recording devices and different implant ages [41]. Qualitatively, the network's spike classifications from two additional subjects were consistent with human expert classifications; quantitatively, the model improved the decoding accuracy compared to traditional threshold-crossing methods [41]. However, the model struggled with diverse spikes from different brain regions and recording devices.

Here, a deep model could be helpful to increase performance. According to early theoretical findings by Cybenko [77], a network equipped with a single hidden layer comprising sigmoid units can effectively approximate any decision boundary if the layer is large enough. However, to avoid an impractically large hidden layer, it is advisable to use deep models. The hierarchical structure of DNNs enables capturing both low-level and high-level features, making them particularly adept at handling variable spikes, superpositions and transient noise. In 2021, Valencia and Alimohammad [42] developed a binarised NN (BNN) for spike classification. In a BNN, weights and activations are binarised to reduce the required memory and computational load. In [42], the BNN takes previously detected and aligned spikes and classifies them according to their underlying neuron. With their SHL-BNN, Valencia and Alimohammad reached an average classification accuracy of 90% on the widely used synthetic dataset that was initially presented in [44]. In sections 4 and 5, it will be seen that full-precision DNNs [78] can achieve an accuracy of more than 99% on the same data, by also using supervised learning schemes. However, the model type and architecture always determine a trade-off between efficiency and effectiveness. In this paper, DNNs refer to models with more than one hidden layer. MLPs, convolutional NNs (CNNs), recurrent NNs (RNNs), especially long short-term memory (LSTM) networks, autoencoders (AEs), and generative adversarial networks (GANs) are the focus of this work.

2.3. Literature overview

The literature was screened using the databases of *PubMed* and *Google Scholar*. The following keywords were used for the search: ('Spike Sorting' OR 'Spike Detection' OR 'Spike Classification') AND ('NN*' OR 'Deep Learning' OR 'MLP' OR 'MLP' OR 'MLP' OR 'CNN*' OR 'ConvNet*' OR 'CNN' OR 'RNN*' OR 'RNN' OR 'LSTM' OR 'LSTM' OR 'AE*'). Due to the large number of publications found, several criteria were defined to narrow the focus this study. The core subjects of this work are deep learning-based spike detection, feature extraction and classification for extracellular recordings of the brain. Only articles that fulfil the following criteria were considered:

- Published until December 2023.
- Peer-reviewed.
- The model presented in the respective article; * has a deep architecture,

 - * was evaluated using extracellular recordings,
 - * is not a denoising or compression model,
 - * is not based on reinforcement learning,
 - * is not an SNN or neuromorphic model,
 - * is not a spike sorting post-processor.

After all relevant literature was gathered, the references of all collected papers were screened to find additional articles within the scope of this survey. Table 1 provides an overview of twenty-four publications considered in this work. Entries in table 1 between two columns indicate that the respective model was used for both tasks in the spike sorting pipeline. As deep classification networks perform feature extraction in the hidden layers of the model, these models are always situated between the 'feature extraction' and 'clussification/clustering' column in table 1. Figure 2 illustrates the breakdown of the different network types utilised in the examined studies. For clarity, classification networks are not represented as feature extractors in figure 2.

3. Spike detection

In this section, the focus is on DNN-based spike detection models. The models' practical relevance is elucidated, highlighting their strengths and weaknesses and comparing them when feasible. In 2017, Lee et al [79] presented a CNN to detect neural spikes; it is part of the well-known spike sorting algorithm called YASS [79]. The first layers in CNNs are known to capture basic features of the input, e.g., edges in images, while deeper layers capture more complex features, e.g. specific objects in images. A relatively small number of kernels (filters, or weights and biases) are applied to the input with a specified stride, which reduces the chances of overfitting and helps to concentrate on specific patterns, regardless of their position in the input. The CNN in [79] takes data from a single channel, propagates it through two hidden layers and returns binary labels ('spike' or 'no spike'). Lee et al used the rectified linear unit (ReLU) activation function, $\sigma(x) = x$ if $x \ge 0$, else 0, to add non-linearity to the system [79]. Moreover, L2 regularisation, also known as weight decay, was applied to prevent overfitting. Tested on various noise levels, the proposed CNN outperformed several contemporary detection techniques [79]. In 2020, a further developed model of YASS was published [118], still using a CNN for spike detection, but processing multi-channel data. Note that [118] has not been peer-reviewed. As in [79], data augmentation was utilised to create a training set based on pre-sorted spikes with artificial overlaps at random scales and shifts [118]. Temporal and spatial filters were segregated to enhance network speed, with the initial layer focusing to extract temporal characteristics and the subsequent layer capturing features from adjacent spatial channels [118]. Lee et al tailored their model to retina recordings, where overlapping spikes occur frequently. 70% of the positive events in the generated training data consisted of spike collisions [118]. YASS [118] consists of the following sub-algorithms: spike detection, denoising, de-duplication, spike division

	Table 1. Re	cently developed spike	sorting solutions usin	g deep learning-based 1	models in chronological	order. Models that are publicly a	vailable are highlighted in bold.
References	Authors	Datasets	Channel type	Detection	Feature extraction	Classification/ clustering	Significance of the deep learning-based approach
[62]	Lee et al	[80, 81]	Single	CNN	PCA	Gaussian Mixture Model	Detection of neural spikes with improved spatiotemporal alignment.
[82]	Yang <i>et al</i>	[44, 83]	Single	Peak detection	PCANet	SVM	Low complexity sorting with similar accuracy to conventional methods.
[84]	Wu et al	[44, 84]	Single	Median-based	0	AN	Few-shot sorting enabled by semi-supervised adversarial representation learning.
[85]	Saif-ur-Rehman <i>et al</i>	[8587]	Single	CNN	1		Universal neural channel tracking that works across species, brain regions & electrodes.
[88]	Rácz et al	[89]	Multi	CNN + LSTM	0	NN	Spatiotemporal spike dassification using HD-MEA data (128 channels).
[78]	Park <i>et al</i>	[2, 44, 90]	Single	Thresholding	N	ſĹ₽	Improved accuracy and runtime compared to standard methods like PCA + k-means.
[91]	Wouters et al	[92]	Multi	1	MLP	k-means	Resolved overlapping spikes in the feature space using a custom cost function.
[93]	Markanday <i>et al</i>	[93]	Single	CNN			Human expert-level detection of complex spikes produced by cerebellar Purkinje cells.
[94]	Ciecierski	[95]	Single	1	0	NA	Unsupervised spike sorting with a GAN, using a 'spike-aware' loss function.
[96]	Eom <i>et al</i>	[2, 44, 90, 97]	Multi	Thresholding	AE	DBSCAN	Unsupervised feature extraction by combining latent space representations of three AEs.
[98]	Li et al	[44, 99]	Single	I	0	NN	Improved spike classification accuracy due to many convolutional layers.
[100]	Saif-ur-Rehman <i>et al</i>	[44, 85–87]	Single	CNN + CNN	PCA	k-means	Background activity rejection reduces the number of waveforms for subsequent analysis.
[101]	Rokai <i>et al</i>	[46, 89, 102]	Multi		AE		Integrated model for online spike detection and classification using HD-MEA data.
[103]	Seong et al	[44, 104]	Single	Thresholding	AE	k-means	Unsupervised feature extraction utilising a quantised model; implemented on an ASIC.
[105]	Yi et al	[2, 105]	Multi	MountainSort	0	NN	Spike classification with a quantised model, using HD-MEA data; FPGA implemented.
[106]	Liu et al	[44, 99]	Single	1	CNN	+ LSTM	Optimised classification of overlapping spikes by combining CNN and LSTM.
[107]	Radmanesh <i>et al</i>	[43, 45, 108]	Single	1	AE	SVM	Noise-robust unsupervised feature extraction using a deep contractive AE.
[109]	Okreghe <i>et al</i>	[44, 86, 87, 110]	Multi	CNN + CNN	PCA	k-means	Improved channel selection and artefact removal using spatiotemporal spike data.
[111]	Ardelean <i>et al</i>	[2, 111]	Single	Thresholding	AE	k-means	Network complexity has to be designed relatively to the complexity in the neural data.
[112]	Rokai <i>et al</i>	[43, 46, 62]	Multi	G	NN	Isosplit5	Input-source agnostic spike detection and feature embedding using edgeTPU hardware.
[113]	Wang <i>et al</i>	[44, 75, 99, 114]	Single	CNN + CNN	CNN	+ LSTM	Spike detection and localisation with 2 CNNs and sorting with a sliding-window LSTM.
[115]	Zhang <i>et al</i>	[44]	Single		0	NN	CNN taking Log-Mel spectrogram as input to dassify overlapping spikes.
[116]	Meyer et al	[44, 99]	Single		MLP		Lightweight detection and sorting with downstream validation to enhance performance.
[117]	Zacharelos et al	[108]	Single		MLP		Lightweight detection and sorting; ASIC implemented, using approximate computing.

IOP Publishing



ŧ	Туре	Dim.	Kernel	Stride
)	Input	32×4		
_	2D-Conv	16 @ 28 × 3	5×2	1×1
2	2D-Conv	$256 @ 24 \times 2$	5×2	1×1
;	Pooling	256 @ 12	2×2	2×2
Ł	LSTM	512	_	_
5	Dense	512	_	—
5	Dense	256	_	_
7	Dense	2	_	_

 Table 2. Model architecture of the CNN-LSTM in [88].

for parallel processing, clustering by using iterative featurisation, triaging and deconvolution. A clear advantage of such a procedural pipeline is that the CNN could be easily replaced with other detection models. The performance of YASS [118] was compared with other state-of-the-art sorters like Kilosort [21], SpyKING CIRCUS [62] and MountainSort [59], where Kilosort achieved the highest performance. YASS was significantly better than Kilosort at restoring spikes on retinal ganglion cell types, as the CNN in [118] was trained especially on retinal recordings. To avoid a potential bias in their evaluation, Lee et al [118] tested their model using a second hybrid dataset, where YASS performed best. The CNN contributes by finding many spikes missed by other sorters and improving spatiotemporal alignment [79].

In 2019, Rácz et al [88] proposed a convolutional LSTM network to detect spikes in HD-MEA data. An LSTM network represents a form of RNN that addresses the intricacies of sequence prediction problems. During a forward pass, it takes a series of inputs and processes each element by maintaining a cell state and a hidden state throughout the sequence. An integral part of its functionality lies in implementing distinctive gating mechanisms, namely the forget, input, and output gates, acting as guardians governing the flow of information. LSTM networks have the ability to capture long-term dependencies within the data through the coordinated interplay of these gates. In [88], recordings from 128 channels arranged in a 32×4 grid were applied to an LSTM network. Table 2 details the model architecture employed in [88] including the type of each layer, data dimensions (Dim.), kernel sizes, and strides. Convolutional layer dimensions are denoted as 'number of filters @ dimensions'. The network was trained using pseudolabels obtained by Kilosort [21]. For regularisation [88], used an early stopping criterion that terminates model training if the loss of the system does not decrease for a specified number of epochs. Another regularisation technique used in [88] that also speeds up training is batch normalisation [119], which forces the layer inputs to have a mean of zero and a standard deviation of one. Rácz et al utilised nine 45 min recordings, using 90% of the data for training and the remaining 10% for testing. Ten positive predictions

were considered as spikes for evaluation [88]. It should be noted that the results of the model in [88] were evaluated in relation to Kilosort's performance, which does not make the achieved results entirely conclusive. However, this is often the case for models that were evaluated using real-world data. The DNN in [88] yielded a true positive rate (also called recall or sensitivity) of around 70% and a precision (accuracy of the model in classifying a sample as positive) of only 19%. In this regard, Rácz et al concluded that low precision should not lead to trouble for sophisticated sorters as they may discard false positives [88]. Nevertheless, this leads to redundant data processing of noisy segments. Furthermore, a recall of 70% still implies a lot of false negatives, making the model in [88] unsuitable for practical use despite extensive training and a sophisticated architecture.

In 2023, Wang et al [113] proposed a method for spike detection working with single-channel data that is based on a two-stage procedure involving two CNNs. Firstly, the continuous single-channel signal was divided into sliding windows of 200 samples, using an overlap of 20%. This segmented data was then applied to a CNN that predicts the input as neural or background activity. The architecture of the CNN in [113] is described in table 3, where $\sigma(x)$ denotes the used activation functions. Note that missing entries for both dense layers do not necessarily imply that no activation function was used but mean that it was not mentioned in the respective paper. As can be seen in table 3, the Softmax activation function, $\sigma(x_i) = e^{x_i} / \sum_j e^{x_j}$, where *i* represents the index of the current class and j denotes the indices over all classes, was utilised to determine probabilities for both classes ('spike' or 'no spike'). In the second stage, spikes were localised within the 200 samplelong signal snippets. The architecture of this model is identical to the one described in table 3, but skips the first convolutional layer. Moreover, this model's final layer predicts the spikes' start and end positions [113]. Both CNNs were trained with labelled data and compared against different network architectures, e.g. LSTM, with the proposed CNN showing the best performance [113]. Experiments with varying noise levels showed that this factor directly influenced the detection quality. However, it is assumed that this is

Table 3. Model architecture of the CNN in [113].

#	Туре	Dim.	$\sigma(x)$	Kernel	Stride
0	Input	200	_		
1	1D-Conv	20 @ 200	ReLU	3	1
2	Pooling	20 @ 50	_	4	4
3	1D-Conv	50 @ 50	ReLU	3	1
4	Pooling	50@12	_	4	4
5	Dense	550	_		_
6	Dense	200	_	_	_
7	Dense	2	Softmax		

Table 4. Model architecture of Spikedeeptector [85].

#	Туре	Dim.	$\sigma(x)$	Kernel	Stride
0	Input	48×20	_	_	
1	2D-Conv	$25 @ 48 \times 20$	ReLU	3×1	1×1
2	2D-Conv	25@46	ReLU	3×20	1×1
3	Pooling	25@23	_	2×1	1×1
4	2D-Conv	50 @ 21	ReLU	3×1	1×1
5	Pooling	50@11	_	2×1	1×1
6	2D-Conv	100 @ 9	ReLU	3×1	1×1
7	Pooling	100@5	_	2×1	1×1
8	Dense	100	ReLU		
9	Dense	2	Softmax	_	

the case for most models presented in this survey, as high noise aggravates distinguishing spikes from artefacts. Irrespective of this, using sliding windows for spike detection ensures that any spike signal intercepted by a window slice will still be captured within the subsequent window, but also introduces redundant data processing. The CNN in [113] is limited to detecting one spike at a time. For practical applications, it is therefore essential to expand this model into a multi-target detector.

3.1. Neural channel selection and artefact removal

In 2019, Saif-ur-Rehman et al proposed Spikedeeptector [85], a CNN designed to detect and track channels containing neural data. This is helpful in identifying useful channels for BCI applications [85]. The proposed CNN is a supervised learning model and relies, like the models described above, on the quality of labelled training data. Saif-ur-Rehman et al used a semi-automatic method to acquire pseudo-labels, including visual inspections [85]. The training data was taken from a single subject (1.56 million feature vectors that are batches of waveforms from a single channel with a size of 48×20 samples). If at least one of these waveforms represented a spike, the whole batch was classified as neural. The CNN applied for this task is detailed in table 4. While the first layer performs a convolution through time, the second layer convolves over the entire batch. For training, the authors used L2-regularisation, early stopping, batch normalisation and dropout [120] as regularisation techniques to improve generalisation. Dropout is a simple yet

effective technique to prevent overfitting by randomly ignoring selected units during training, preventing certain co-adaptations on the training data. As the CNN in [85] only assigns labels to a given feature vector, Saif-ur-Rehman introduced a criterion that calculates the mode of the predicted outputs to classify entire channels as 'neural' or 'noisy'. Overall, an accuracy of 97.2% could be achieved with Spikedeeptector [85]. Notably, the model could generalise across different brain regions, subjects, species and electrode types. Spikedeeptector may be used to identify and exclude threshold-crossing artefacts from subsequent analysis. However, complex artefacts can still affect the model as these events may occasionally be classified as neural. In a follow-up study published in 2021 [100], Saif-ur-Rehman et al developed an additional CNN called 'background activity rejector' (BAR) that takes signals selected by Spikedeeptector and discards background activity by classifying a given signal as spike or noise [100]. The architecture of BAR is detailed in table 5. This model also utilises batch normalisation and L2 regularisation to prevent overfitting. Labelled data of two species, five subjects, six brain areas, three different types of electrodes and two recording systems were used for training to create a robust system [100]. Evaluated on data from three human subjects, BAR could achieve an accuracy of 92.3%. Further considerations regarding the scalability of this model can be found in section 7.1.

In 2023, Okreghe *et al* [109] also worked on a solution for improved channel selection and artefact removal. The basic approaches of Saif-ur-Rehman

#	Туре	Dim.	$\sigma(x)$	Kernel	Stride
0	Input	48			
1	1D-Conv	25@48	ReLU	3	1
2	1D-Conv	50@44	ReLU	5	1
3	Pooling	50@22	_	2	1
4	1D-Conv	50@18	ReLU	5	1
5	Pooling	50@9	_	2	1
6	Dense	50	ReLU		
7	Dense	2	Softmax	_	_

 Table 5. Model architecture of BAR [100].



Figure 3. Spike detection method from Saif-ur-Rehman *et al* [100] and Okreghe *et al* [109]. The first CNN was employed to detect neural channels. Okreghe *et al* used multi-channel data as an input to their CNN while Saif-ur-Rehman *et al* applied their model to single-channel data. The next step involved applying a second CNN to the chosen neural channels in order to identify neural events and remove noise and artefacts. The identified neural events were then used for further analysis.

et al [100] and Okreghe et al [109] are similar and illustrated in figure 3. The method in [109], called 'deep spike detection' (DSD), also works with two CNNs: one for the selection of channels and one for artefact removal. Like the work above, DSD was trained with many labelled feature vectors (1.61 million) to create a robust network. Contrary to the work of Saif-ur-Rehman et al [85], the authors of [109] applied Kilosort [21] to obtain pseudo-labels. Moreover, instead of taking twenty waveforms from a single channel, they used waveforms from multiple channels as an input for DSD to utilise spatiotemporal data dependencies, similar to [88, 118]. The first CNN has the same architecture as Spikedeeptector (table 4). As in [85], a batch (48×20) was labelled as 'neural' if it contained at least one spike waveform [109]. Naturally, a larger batch size increases inference time, which may affect online processing, but also yields better results (99.62% for a batch size of 100; 97.51% for a batch size of 20). Like BAR [100], the second CNN (artefact removal network) is provided with a single feature vector (48 \times 1) and discards non-neural events [109]. The architecture of this CNN also resembles BAR, which was presented in [100] (table 5). Using the data in [109], the artefact removal network could achieve an overall accuracy of 92.3%. Using both simulated and experimental datasets, integrating DSD into a traditional

spike sorting pipeline (PCA + k-means), as done in [100], demonstrated a respectable level of sorting [109]. In contrast to the work of Saif-ur-Rehman *et al* [100], DSD was tailored to a specific class of artefacts, making it less generic and more responsive to deep temporal spikes [109]. Despite the similarity of the models from Saif-ur-Rehman *et al* and Okreghe *et al*, they serve slightly different purposes. While Spikedeeptector and BAR were built to track neural channels that contain useful data for BCI applications and discard background activity in the second step, the solution from Okreghe *et al* screens multiple channels for neural activity and discards artefacts.

3.2. Detection of complex spikes (CSs)

Ideally, spike detection models can detect spikes at different scales and even under adverse conditions. However, Purkinje cells produce 'CSs' that show high misclassification rates with conventional algorithms due to their polymorphic complexity [93]. The role of CSs has been controversially discussed in the past. While it has been hypothesised that CSs play a significant role in motor timing [121] or performancebased motor learning [122], not all experiments in the past were conclusive in this regard [93]. Muller et al recently found that CSs modulate the simple spike output of Purkinje cells, which influences certain behaviour, such as eye saccades [123]. Since CSs could only be reliably detected manually by experts for many years, Markanday et al [93] developed a CNN for this purpose. Compared to usual spikes, CSs are rare events whose waveforms exhibit specific characteristics, as they are relatively long-lasting, polyphasic events, often showing inconsistency during a single recording. In [93], a human expert labelled the start and end points of CSs in 159 recorded Purkinje cells with 24 segments per cell, each with a length of 25 0 ms [93]. The famous U-Net [124] inspired the architecture of the utilised CNN. Given the reliability with which human experts reach a consensus on the presence of CSs, the input of the CNN was twofold: (i) band-pass filtered spikes with a frequency range of 300 Hz to 3 kHz; (ii) LFP with a frequency band of 30 Hz to 400 Hz [93]. With a kernel size of nine, they used by far the largest kernel of all CNNs described

in this work, as CSs are longer events (~ 10 ms) than usual spikes ($\sim 1-2$ ms) [93]. The method in [93] includes a three-step post-processing procedure to improve the system's overall performance. The first step realigns the detected CSs. The second step maps spikes and LFP waveforms occurring within 2 ms after the CS start into a two-dimensional feature space using the uniform manifold approximation and projection algorithm [125]. Subsequently, DBSCAN was employed to identify spike candidates. Using this technique, usual spikes could be separated from CSs [93]. Combined with downstream post-processing, the model in [93] proved superior to a popular spike sorter and a PCA-based algorithm for CS detection, despite being trained with little data. Another method that was frequently applied in the past to detect CSs is the template matching-based alpha omega engineering multi spike detector (MSD) [126]. However, its performance was limited due to the inconsistent waveform characteristics of CSs [93]. According to Markanday et al, their method outperformed MSD [93].

4. Feature extraction

In 2017, Yang et al proposed an approach to extract neural features with PCANet [82], a DNN that utilises the basic principles from PCA. While PCA is a linear dimensionality reduction technique, a deep learningbased PCANet applies non-linearity across its layers. Features can be learned hierarchically and extracted locally in PCANet [82]. Moreover, PCANet runs faster than traditional PCA as it avoids the computationally demanding eigen-decomposition by applying filters directly to a given input. Yang et al considered these advantages to build a faster model that utilises PCA principles and overcomes the classical downsides of PCA [82]. Using conventional PCA, the extracted spikes and their corresponding eigenvalues were clustered by employing the k-means algorithm [82]. The cluster centres were then taken as spike templates and were shifted to create a Toeplitz matrix where the columns were used to train the PCANet [82]. In the subsequent stage, the resulting eigenvectors of PCANet were utilised as inputs for an SVM to classify spikes according to their underlying neuron. Following Yang et al [82], the proposed method had lower complexity than conventional methods, like PCA and k-means, but indicated no superior performance.

In 2020, Wouters *et al* [91] proposed a feature extraction method to resolve overlapping spikes in the feature space, a problem that often limits clustering methods in separating spikes from distinct neurons. In [91], an MLP was trained using the custom cost function in equation (1),

$$\sum_{i,j\in T} g \left\| (\boldsymbol{s}_i) + g \left(\boldsymbol{s}_j \right) - g \left(\boldsymbol{s}_i \oplus \boldsymbol{s}_j \right) \right\|_2^2 + \frac{\kappa}{\left\| g \left(\boldsymbol{s}_i \right) - g \left(\boldsymbol{s}_j \right) \right\|_2^2}$$
(1)

where g corresponds to the non-linear feature map, and $s_i \oplus s_j$ denotes the superposition of spikes from two different neurons *i* and *j*. The hyperparameter κ was determined by a grid search and set to ten [91]. The first component of the cost function decreases by treating features as a linear operator on a linear combination, while the second term enhances the separation between spikes of distinct neurons in the feature space, thereby avoiding convergence to trivial solutions [91]. The MLP in [91] takes 300 features (30 samples \times 10 spikes) as input and processes the data through two hidden layers with 600 and 60 units. The output layer has three units, and all layers use the ReLU activation function. The training set consisted of 50 000 waveforms, where three samples were created for each pair of neurons i and *j*: s_i , s_j , and $s_i \oplus s_j$ [91]. To create s_i and s_j , random white noise was superimposed with artificial spikes, aligning them based on their highest peak and adjusting the time dimension to a specified window size of 30 samples. Creating $s_i \oplus s_i$ involved randomly shifting the template of neuron *i* in relation to the spike template of neuron *j* [91]. Wouters et al could demonstrate that the trained MLP in [91] reliably resolves spike overlaps in the feature space, thereby improving subsequent clustering with k-means. This method and other models, which were dedicated to overlapping spikes, are revisited in section 7.3.

AEs are powerful tools for spike-denoising [127], data compression [128], and feature extraction, as we will see in this section. Basic architectures of AEs with varying depths can be seen in figure 4. The input signal is processed by the encoder module (layers with gradually decreasing size in figure 4) until it passes the coding layer (also called bottleneck; layers with three units in the middle of the AEs in figure 4) and is propagated through the decoder module that tries to reconstruct the input signal (bottom parts of the AEs in figure 4). As the encoder drastically reduces dimensions, the model creates a low-dimensional latent space that often facilitates clustering. In 2023, Ardelean et al [111] analysed different AE architectures, including shallow AEs, deep AEs, and LSTM AEs. The authors of [111] used synthetic data [2] (single-channel recordings with up to 20 active neurons) and experimental recordings [111] in their experiments. Among all architectures evaluated by Ardelean et al, the shallow AE performed best on the synthetic data, which can be attributed to the low data complexity (high SNR and no spike overlaps) [111]. This finding is consistent with standard deep learning principles, as the size (and layer complexity) of the



hidden space should always be adjusted to the complexity of the problem [129]. Using the experimental data, this could be supported as the AE with eight hidden layers in the encoder and decoder achieved better results than the shallow AE [111]. Ardelean *et al* demonstrated that their model isolates 10 out of 17 distinct spike clusters in a specific synthetic recording [111]. However, other studies reveal that spike data with such 'low complexity' can be sorted precisely with other unsupervised machine learningbased methods, as Nadian *et al* [130] utilised t-SNE in combination with DBSCAN, achieving an accuracy of >90% on the synthetic data in [2] with up to 20 active neurons.

In 2020, Eom et al [96] presented a method to extract features with an ensemble of AEs based on the idea that models of different dimensions capture different spike characteristics. They detected spikes with amplitude thresholding [44], followed by obtaining the spikes' gradient values, which is advantageous in signal processing tasks [131]. Subsequently, the spikes were temporally aligned to facilitate feature extraction. Eom et al found that shallow networks capture general spike attributes, while deeper networks better represent details of a given spike [96]. Hence, they applied the pre-processed signals to three differently sized AEs, which are shown in figure 4. The ReLU activation function was employed to add non-linearity to the hidden layers. The hidden codes of each model were concatenated to generate a nine-dimensional feature space, which was clustered using DBSCAN [96], as illustrated in figure 4. Eom et al achieved an accuracy of 99.81% on the widelyused synthetic data in [44] (overlapping spikes were excluded for this evaluation). Furthermore, multidimensional input (concatenated Tetrode-data) in a one-dimensional input vector improved the sorting accuracy from 96.58% to 98.27%, as the model could extract more valuable features [96]. Further considerations regarding the scalability of this and the following model can be found in section 7.1.

In 2021, Seong *et al* [103] also proposed an AE for feature extraction. Their method includes the

following steps: detection, alignment, initial template generation, feature extraction, and clustering. To detect spikes, the dual vertex threshold method [132] was employed. Then, the initial spike templates were generated by using the first 100 aligned spikes in the data, and the number of centroids for clustering was determined [103]. The AE in [103] is detailed in table 6. The decoder network has a mirrored architecture to the encoder. Notably, L2normalisation was applied to the last layer of the encoder as this facilitated clustering. Moreover, a 'spike aware' loss function was designed to assign more importance to the middle of spikes (peak and valley samples). For clustering, a k-means algorithm based on cosine similarity that is robust against false positives was utilised [103]. Using the synthetic data in [44], Seong et al reached an accuracy of 95.54%. It has to be considered that Seong et al evaluated their model using 5-bit weights and 7-bit activations, as the goal in [103] was to build an efficient hardware solution, which is described in section 7.2. As already observed in some detection models, performance in high-noise environments remains challenging [103], as a low SNR leads to stronger signal corruption, which increases the intraclass variance of spike features and thus complicates clustering.

To address this problem, Radmanesh et al [107] built contractive AEs (CAEs), which include a contractive penalty term in the loss function, that penalises the Frobenius norm of the Jacobian matrix of the encoder activations, forcing the learned representations to exhibit invariance to minor variations in the input. The model architectures in [107] were optimised on each recording individually by using a gridsearch algorithm. The smallest model applied to the data in [44] had a dense architecture with 72, 45, 20 and 3 units in each layer, while the biggest model, used for the data in [45] was equipped with 90, 70, 35 and 10 units, respectively [107]. Although this technique yields optimal results, it may also lead to models that are overly adapted. In contrast to [96] and [103], the authors of [107] utilised labels derived from a k-means algorithm to train an SVM for classification. Radmanesh et al outperformed several state-ofthe-art methods, such as SPC [44], IronClust [63], and MountainSort4 [60] by achieving accuracies of 97% and 96% on two specific recordings from [44], 91% on the data in [108], and 87% on a recording from [45]—all while maintaining a faster processing speed [107]. Notably, the CAEs showed minimal sensitivity to the size of the training set, making them good candidates for even small datasets and online inference [107]. The benefit of using contractive loss could be further demonstrated, as simple AEs without contractiveness could not compete with the used CAEs [107].

Table 6. Model architecture of the encoder network in [103].

#	Туре	Dim.	$\sigma(x)$	Kernel	Stride
0	Input	50		_	
1	1D-Conv	32@25	ReLU	3	2
2	1D-Conv	64@13	ReLU	3	2
3	1D-Conv	96@7	ReLU	3	2
4	Dense	4	_		_



Figure 5. Spike sorting strategy proposed by Park *et al* [78]. Offline training: after band-pass filtering of the signal, spikes get detected and aligned. PCA and k-means are used for feature extraction and clustering. Subsequently the MLP is trained with the spikes that are closest to the centroids in the feature space to minimise labelling errors. Online mode: after being trained, the MLP can classify detected and aligned spikes.

5. Spike classification

This section presents straightforward classification models and more advanced solutions developed to address some of the mentioned challenges in spike sorting, such as overlapping spikes or limited prior information about the data. Supervised NNs require input (spike) and output (label) combinations to learn underlying data patterns but usually achieve higher accuracy than unsupervised models. Both methods can be combined, where unsupervised models are used during an initialisation phase to label a training set for the supervised model that can be trained subsequently to perform online spike sorting. This was also done for the spike detection models in [109] and [88], with Kilosort [21] as the utilised tool to generate binary pseudo-labels ('spike' or 'no spike'). However, to sort spikes according to their underlying neuron, the spike IDs, e.g. 'spike A', 'spike B' and 'spike C', must be obtained. In 2019, Park et al [78] utilised an MLP for spike classification. The strategy in [78] is illustrated in figure 5. Park et al [78] used PCA and k-means to create pseudo-labels, using only the spikes in the data that were located closest to the centroids in the feature space to minimise labelling errors [78]. After training, the MLP assigns each spike with a label by using a Softmax layer with n units for n possible classes. The MLP in [78] has four hidden layers with 256 units each that use the hyperbolic tangent activation function, $\sigma(x) = 2/[(1 + e^{-2x}) - 1]$. The results demonstrate that unsupervised pseudo-labelling with PCA and kmeans combined with subsequent supervised classification yields better performance than only applying PCA and k-means or SPC [78]. However, the unsupervised training strategy limits the model's classification performance. Accuracy increased from 51.55% to 99.55% on a specific recording when trained with the actual labels [78]. Thus, a change towards a more advanced unsupervised labelling strategy may result in fewer erroneous labels and therefore could improve performance. The scalability of this and some of the following approaches are discussed in section 7.1.

In 2020, Li et al [98] also worked on the classification of spike waveforms. For the synthetic data [44], Li et al used the true labels available with the dataset to train their model, varying the size of the training set between 5% and 50%. However, they did not mention spike alignment, even though this step is of big importance as the spikes have cluster-specific shifts with respect to the given ground-truth information. If these shifts are not corrected by an alignment procedure, there is a risk that the respective deep learning model learns these characteristic peak positions, which are not present in real scenarios. Hence, the model in [98] could be biased in this regard. The architecture of the CNN in [98] is summarised in table 7. Li et al regularised their model using batch normalisation and dropout. Evaluated on the data in [44], the CNN [98] achieved high accuracy, performing slightly better than the MLP in [78] (even when trained with the true labels), at the expense of higher computation. However, depending on whether the spikes were aligned before model training or not, the possible bias mentioned above may have led to falsely promising results. Naturally, more training data increased performance. Nevertheless, even with 170 labelled spikes (5% of the available data), the CNN achieved >99% accuracy for most synthetic signals [98]. Using real extracellular data [98], the average accuracy of this CNN dropped to 96.53%, which may be due to inherent challenges of real data, such as class imbalance and more realistic noise. Moreover, some recordings in [98] do not contain enough data from a single neuron to train the model [98]. Using the experimental data, the CNN

#	Туре	Dim.	$\sigma(x)$	Kernel	Stride
0	Input	64	_		
1	1D-Conv	32@64	ReLU	3	1
2	1D-Conv	64@64	ReLU	3	1
3	Pooling	64 @ 32	_	2	2
4	1D-Conv	128@32	ReLU	3	1
5	Pooling	128 @ 16	_	2	2
6	1D-Conv	128@16	ReLU	3	1
7	Dense	300	ReLU		
8	Dense	100	ReLU		
9	Dense	n	Softmax	—	

 Table 7. Architecture of the CNN in [98].

Table 8. Model architecture	of the LSTM-CNN in [113].
-----------------------------	---------------------------

#	Туре	Dim.	$\sigma(x)$	Kernel	Stride
0	Input	10×7	_	_	
1	LSTM	7			_
2	2D-Conv	25@7×7	ReLU	3×3	1×1
3	2D-Conv	50@7×7	ReLU	1×1	1×1
4	Dense	50	_	_	
5	Dense	25	_	_	
6	Dense	п	Softmax	_	

in [74] was trained with spikes that were pre-sorted using SPC and manual re-sorting [44]. Therefore, the proposed method resembles the one in [78], as an unsupervised method was used to create pseudolabels. Note that for the data in [98], spikes were aligned before pre-sorting with SPC.

In 2023, Wang et al [113] proposed a convolutional LSTM for spike classification. The detection unit, consisting of two CNNs, was explained in section 3. The output of the second CNN defines the start and end points of a potential spike with a length of 40 samples. The authors of [113] split each predicted waveform (40×1) into seven sliding windows (10 \times 7) with an overlap of five samples for each window. This was then applied to an LSTM layer that created seven time-series features. These features were spliced into a 2D-feature matrix (7×7) to construct the input for the subsequent convolutional layers [113]. The architecture of the model in [113] is shown in table 8. Comparing the results from Li et al [98] and Wang et al [113] on the synthetic data [44] reveals that both models performed similarly. Li et al [98] reached 99.08% accuracy using 5% of the data for training, while Wang et al [113] achieved 98.85%; using 40% of the data for training, they reached an accuracy of 99.58% and 99.55%, respectively. By processing ten individual channels of the experimental data in [99] containing the activity of three, four and five neurons, Wang et al reached an average accuracy of 94.77% with 50% of the available data used for training, whereas the CNN from Li et al [98] reached 95.66% on the same channels.

5.1. Classification of overlapping spikes

Recent studies, like the one by Liu *et al* [106] in 2022, or the one by Zhang et al [115] in 2023, focused explicitly on the problem of overlapping spikes. Both of these studies also did not mention spike alignment, which may introduce the same bias as mentioned above. The model used in [106] offers a combination of CNN and LSTM, which is detailed in table 9. Batch normalisation was utilised for the convolutional layers to regularise the model. Notably, a Softmax layer with m units was utilised to classify the input: one class for each spike group and additional units for all possible overlapping combinations of spikes from two neurons. The model in [106] could distinguish between these classes as it was trained with augmented data consisting of single spikes and artificial overlaps. Compared with the study in [98], results could be improved, especially on spike collisions [106]. The proposed technique to address overlapping spikes is discussed in section 7.3.

Zhang *et al* [115] addressed the spike classification problem using the Log-Mel spectrogram and a CNN. Spectrograms are widely used in signal processing, particularly for speech recognition and music analysis. Furthermore, logarithmic scaling in the spectrogram enhances its informativeness and adaptability for various signal-processing applications. The procedure in Zhang *et al* [115] has two parts. The first part utilises the short-time Fourier transformation to acquire the signal's spectrogram, which was converted into a Log-Mel spectrogram to reduce the amount of data while extracting advantageous features [115]. In the second part, a CNN is utilised to process the

Table 9. Model architecture of the LSTM-CNN in [106].

#	Туре	Dim.	$\sigma(x)$	Kernel	Stride
0	Input	64			
1	1D-Conv	32@64	ReLU	3	1
2	1D-Conv	32@64	ReLU	3	1
3	Pooling	32@32		2	2
4	LSTM	n/s			
5	Dense	300	ReLU		
6	Dense	100	ReLU		_
7	Dense	т	Softmax	_	—

Table 10. Model architecture of the CNN in [88].

#	Туре	Dim.	$\sigma(x)$	Kernel	Stride
0	Input	$32 \times 4 \times 6$	_	_	
1	2D-Conv	n@32 × 4	Linear	$4 \times 2 \times 6$	$1 \times 1 \times 1$
2	Dense	n	Linear	_	_
3	Dense	n	Softmax	—	—

spectrogram. This CNN contains four convolutional layers with 2×64 and 2×128 filters, two average pooling layers, and two dense layers. As in [106], a Softmax layer predicts one of *m* classes [115]. This was also achieved by training the model on both single-spike waveforms and artificial overlaps. The size of the training set is not specified in this work. Using the data in [44], Zhang *et al* reached an accuracy of 95.74% [115], whereas Liu *et al* [106] reached more than 99.5%, suggesting that the use of spectrograms may not provide additional benefits for deep learning-based spike sorting. A similar finding comes from Ardelean *et al*, who also demonstrated that spectral analysis does not offer an advantage in capturing distinct spike features [111].

5.2. Spatiotemporal classification models

Rácz et al [88] proposed CNNs using 2D- and 3Dinput to sort spikes obtained with a 128-channel HD-MEA. Figure 6(a) illustrates how spikes of different neurons may appear spatially on a 32×4 electrode grid at their negative amplitude peak. Figure 6(b) shows how a spike evolves in a spatiotemporal way and how these patterns can be used for signal analysis. The model in [88] that utilises 2D input only took spatial data as input, with spikes at their amplitude peak (cf figure 6(a)). More precisely, frames with an offset of 1-2 samples were used, as these frames yielded the best results [88]. A model with 3D input was also created, using six frames per spike, where each adjacent frame was five samples apart to keep the training time low [88]. The architecture of the CNN in [88], using 3D input and assuming there are n actively firing neurons in the recording, is displayed in table 10. For the data used [89], n ranged from 23 to 46, determined with Kilosort [21], which was also utilised to generate pseudolabels for the supervised models. The results indicate that the 3D-model provides higher accuracy than the 2D-model, as unique temporal spike characteristics provide crucial information [88]. As for the synthetic data mentioned above, pre-sorting with Kilosort also resulted in cluster-specific shifts, that should have been corrected by alignment before network training. However, this was not taken into account in this study either, which also potentially skewed the results. Nevertheless, the method of Rácz et al shows significant advantages over the aforementioned classification models, as the CNN in [88] processes many channels at a time by using a compact network architecture. Another issue was identified in this study: even though classes are imbalanced, the examples per class were not rebalanced for training, which can lead to a deep learning model that weights frequently occurring classes more heavily than examples of rare classes. This was also neglected by the studies mentioned above that were using experimental data [98, 106, 113]. Rácz et al also investigated the feasibility of spike predictions before they appear in the signal. This would be useful for BCIs as the delay between sorting and actuation could be reduced. It was shown that the initial waveform (not containing the peak voltage) enabled spike classification to a certain extent, indicating that the initial spike frames contain essential information of a unit's identity. However, predictions before the firing of a neuron were not possible [88]. More reflections on the scalability of this and the following model can be found in section 7.1.

In 2022, Yi *et al* [105] also proposed a multichannel spike sorting model. The data were captured using 64 channels, 49 of which were processed in [105]. The utilised recording shows the activity of 27 neurons. MountainSort [59] was used for labelling, which produced data batches of 49×180 samples for all relevant events (180 samples = 6 ms). Yi *et al* [105] reduced the size of spike snippets to 49 channels x 30 timesteps, as this duration was sufficient to capture





	14			, , , , , , , , , , , , , , , , , , ,	
#	Туре	Dim.	$\sigma(x)$	Kernel	Stride
0	Input	$49 \times 6 \times 5$	_	_	_
1	2D-Conv	2@46×2	ReLU	$4 \times 4 \times 5$	$1 \times 2 \times 1$
2	1D-Conv	2@45	ReLU	4	2
3	Dense	27	Softmax	_	_

Table 11. Model architecture of the CNN in [105]

Table 12. Model	architecture	of the end	coder in	[84]
-----------------	--------------	------------	----------	------

#	Туре	Dim.	Kernel	Stride
0	Input	64		
1	1D-Conv	64 @ 64	1	1
2	Residual block	256 @ 64	3	1
3	Residual block	256 @ 64	3	1
4	Dense	512	_	_
5	Dense	512	_	
6.1	Dense (y)	п	_	_
6.2	Dense (z)	d	—	

relevant spike features [105]. Naturally, a smaller input leads to faster processing of the DNN. However, instead of using every fifth sample from each spike waveform, as Rácz et al did in [88], Yi et al processed the entire waveform to avoid losing important spike features. The input waveform was divided into five sub-snippets, leading to an input size of $49 \times 6 \times 5$ per spike. According to Yi et al, this approach prevents the model from overfitting to certain noises and guides it to better focus on learning spatial features [105]. The CNN used in [105] is detailed in table 11. The authors in [105] could attain a testing accuracy of 93.1% employing full-precision weights and 86.1% using quantised weights with 4-bit precision [105]. The hardware implementation of this approach is briefly discussed in section 7.2.

5.3. Spike classification with GANs

In a GAN, adversarial representation learning can be utilised by employing a generator to produce realistic data samples and a discriminator to distinguish between genuine and generated samples. This approach helps in acquiring meaningful data representations and can improve spike sorting quality by extracting robust features from limited labelled data. In 2019, Wu *et al* [84] developed a semi-supervised spike classification method called few-shot spike sorting (FSSS). They presented a semi-automatic threshold-based labelling technique called DidacticSort to create cluster candidates for the spikes in the utilised data. Experiments were obtained with 50, 100 and 200 labelled spikes from the synthetic data in [44]. The encoder (generator network) of FSSS is detailed in table 12. Two residual blocks, which can be seen as skip connections that add the input directly to the output, were employed to avoid vanishing gradients. The encoder has two outputs: (i) an encoded vector *y* to predict one of *n* cluster labels; (ii) a hidden code z (low-level feature representation). As the output of the encoder *q* is twofold, Wu *et al* [84] used two discriminator networks called D_v and D_z , which are two-layer MLPs with 512 units. The GAN was trained following a three-step algorithm, which is illustrated in figure 7. The authors of [84] successfully categorised spikes within the utilised synthetic data: using 50 labelled spikes from a recording with noise a standard deviation of 5%, an accuracy of 97.7% was reached; by increasing the number of labelled spikes to 200, the accuracy improved to 98.2%. In the presence of noise with a standard deviation of 20%, they reached accuracy levels of 66.4% (50 labelled spikes) and 85.5% (200 labelled spikes) [84]. Also tested on



Figure 7. Training procedure of the model in [84]. (1) The encoder q and the decoder p are trained to reconstruct given spikes by minimising the reconstruction loss L_{recon} , where x and x' are the unlabelled spikes and their reconstructed versions, respectively. (2) The discriminators D_y and D_z are trained to distinguish the generated y and z from samplings of the categorical distribution y_c and d-dimensional Gaussian distribution z_g by adjusting the model weights according to L_{cat} and L_{gauss} . Furthermore, q is trained to better fool D_y and D_z by minimising the regularisation loss L_{reg} . (3) The supervised part comes last, when q is updated by minimising the cross-entropy loss L_{CE} on a mini-batch of labelled spikes \hat{x} , where the labels \hat{y} were obtained by DidacticSort.

experimental data, it was demonstrated that training the GAN on a particular waveform with only a 'few shots' was sufficient to recognise similar spikes in inference mode. In the following, it is discussed that a GAN approach can also be carried out in a fully unsupervised way.

In 2020, Ciecierski [94] proposed a similar method for spike sorting that also relies on the interplay between a generator network and two discriminators. Similar to [84], pre-processing in [94] involved spike detection and alignment to facilitate sorting with the used model. The encoder in [94] processes the input through two dense layers with 100 units each, before the model splits into a categorical output and a Gaussian output. Ten units were used for the categorical output, as it was assumed that no more than ten distinct neurons were firing in the respective recording. Notably, this number represents an upper limit and does not determine the number of active neurons in the recording [94]. Three units were chosen for the Gaussian output. For the first two hidden layers, the author in [94] applied the exponential linear unit activation function, $\sigma(x) =$ x if $x \ge 0$, else $\alpha (e^x - 1)$, with α as a hyperparameter. The decoder was designed the same way as the encoder but with an additional hidden layer after both inputs. Furthermore, Ciecierski used a 'spike aware' loss function, similar to [103]. The ability to enforce both desired distributions (categorical and Gaussian) in the latent space was achieved by using discriminator networks during training, similar to the work of Wu et al [84]. Both the categorical and the Gaussian discriminators were designed as 3-layer MLPs with 100 units in the hidden layers [94]. The training setup for the proposed GAN is similar to the work in [84], but has no supervised part. The model in [94] could identify nine distinct spike waveforms during training, validation and testing, using experimentally obtained recordings [95]. Unfortunately, this model

was not evaluated on the widely-used synthetic data in [44], and thus the performance of FSSS [84] cannot be compared with Ciecierski's model.

6. Integrated systems

Recently, Meyer *et al* [116] and Zacharelos *et al* [117] proposed lightweight MLPs for simultaneous spike detection and spike classification using single-channel recordings and supervised learning. Also, Rokai *et al* [101, 112] presented a mixture of supervised and unsupervised, computationally more intensive solutions, for integrated multichannel spike sorting by employing more complex network configurations.

6.1. Integrated systems for single-channel recordings

In 2023, Meyer et al [116] proposed DualSort, which was trained using synthetic and real extracellular recordings, labelled with the spikes' IDs and timings. Data augmentation with noise was employed to create an extensive training set, similar to the work in [118]. Furthermore, the spikes were extended by shifting them along the time axis in order to train the model to become shift invariant. This enabled classifying spikes at multiple adjacent time steps [116]. The architecture of this model is detailed in table 13. The first output layer was used for spike detection and localisation at twenty different positions in the input, indicating the exact spike timing, while the second output layer predicts one of *n* spike classes [116]. Both output layers contain an additional unit to indicate when no spike is present [116]. Dropout, batch normalisation and early stopping were utilised to regularise the model. Meyer et al also designed a post-processing algorithm that verifies the output of the MLP by validating whether the refractory period of individual neurons was violated, and by calculating the mode of the last 20 predictions to only keep the most frequent classification [116]. DualSort offers high accuracy (98.04% with post-processing; 97.34% as a singleshot model) on the widely-used synthetic dataset [44] outperforming other advanced spike sorting models like WMsorting [133]. The spike times could be predicted precisely as well (84.07% of all predictions were exact; wrong predictions had a mean deviation of 1.1 samples @ 24 kHz). Misses in this regard were mainly caused by overlapping spikes. Using the experimental recordings in [98], it was even possible to outperform the CNN from Li et al [98], especially when trained with little data. The post-processing algorithm helped to increase the robustness against misclassifications, as these often stood out as outliers in larger prediction sequences [116]. The scalability of this and other integrated systems is discussed in section 7.1.

Later in 2023, Zacharelos *et al* [117] also presented an MLP for integrated spike detection and classification, which has many similarities with DualSort

#	Туре	Units	$\sigma(x)$
0	Input	25	
1	Dense	100	ReLU
2	Dense	100	ReLU
3.1	Dense	21	Softmax
3.2	Dense	n + 1	Softmax

Table 13. Model architecture of the MLP in [116].

[116]. However, the authors of [117] also proposed a hardware implementation, which is briefly described in section 7.2. Neurocube [108] was utilised in [117] to generate synthetic extracellular recordings with three active neurons. Different MLP architectures were investigated in [117], where the optimal solution had two hidden layers with 24 and 16 units. The output layer of the model in [117] resembles the second output layer of DualSort [116], as three units were used to indicate activity from the three firing neurons and one unit was utilised to indicate pure noise activity. Both hidden layers apply the ReLU activation function, and the Hardmax activation function was applied to the output layer [117]. Hardmax, also known as the argmax function, identifies the unit with the highest value, which saves resources compared to Softmax. For regularisation, the authors in [117] used early stopping and L2-regularisation. Tested on the synthesised recordings, Zacharelos et al reached an accuracy of 96%. The model in [117] has a leaner architecture than DualSort but was optimised and tested on a single dataset, which does not allow any conclusions on the generalisation of the model in [117]. Additionally, the model in [117] cannot gain exact information regarding the spike timing. Hence, if the spike trains need to be reconstructed by the network outputs, DualSort has a slight advantage in spike time accuracy.

6.2. Integrated systems for multi-channel recordings

In 2021, Rokai et al [101] presented ELVISort, an integrated multi-channel spike sorting system based on a variational AE (VAE). In VAEs, loss is augmented to force the hidden code to have a normal distribution, which regularises the model and leads to more consistent latent representations. The encoder contains two branches of Bi-LSTM and 2D-convolutional layers with dense layers at the end [101]. A Bi-LSTM operates by processing inputs in forward and backward directions through the hidden layers before the outputs get combined at each time step. The dense layers in ELVISort are followed by three branches. The first branch reconstructs the input and consists of LSTM and attention layers, as this improved the generalisibility of ELVISort [101]. Attention layers allow to selectively focus on specific segments of the input data dynamically, rather than treating all inputs the same way. The second branch is for

classification, which is used to fine-tune the model for spike detection using a Softmax layer with two units ('spike' or 'no spike'), while it was tuned to the number of clusters *n* in the respective recording for sorting (this is the only supervised part of the network). The third branch is for clustering, where Student's t-distribution was utilised to assign soft labels during training and k-means for centroid initialisation. During training, the reconstruction capability of ELVISort was given the highest priority. ELVISort [101] was evaluated on three experimental datasets using the F_1 -Score, a statistical measure used to evaluate the performance of a classification model that calculates the harmonic mean of precision and recall and ranges from zero (minimum) to one (maximum). Using the data in [102], ELVISort achieved an F_1 -Score of 0.964, which is competitive to several state-of-the-art models, such as Kilosort (F1: 0.98) and IronClust (F_1 : 0.97) [101]. Using the hybrid dataset in [46], ELVISort reached an F_1 -Score of 0.81, a comparable result to SpyKING CIRCUS, Kilosort2, and IronClust, all achieving an F_1 -Score of 0.83 [101]. Although the unsupervised part of ELVISort exhibited a commendable capacity for detection and sorting, the cluster-specific accuracies were loosely correlating with the average SNR of the cluster, suggesting that the performance bottleneck could be in the feature extraction part of the algorithm [101].

In 2023, Rokai et al presented a new solution for integrated spike sorting [112], using tensor processing units (TPUs) to accelerate sorting. They developed a semi-automatic procedure where spikes were averaged from a specific cluster to identify the primary channel number, which was verified by a human expert [112]. The model consists of a self-supervised and a supervised model. The selfsupervised model takes 105 features as input and processes the data through three residual blocks, four 1D-convolutional layers and one dense layer at the end, where it outputs 32 features (spike embedding). Batch normalisation was used for regularisation, and the model utilised the leaky ReLU activation function, $\sigma(x) = x$ if $x \ge 0$, else αx , with α as a hyperparameter that controls the slope for negative inputs. Nearestneighbour contrastive learning NNCLR [134] was chosen as the self-supervised method in [112]. The training procedure of the self-supervised method is illustrated and explained in figure 8(a). The model in [112] was trained with multiple recordings at once



Figure 8. Training scheme of the model from Rokai et al [112]. (a) The encoder of the self-supervised model takes an instance spike and a cluster mean spike as input and creates two different feature embeddings. These embeddings are then processed by a projection head. By applying NNCLR, the contrastive loss is calculated to pull embeddings from positive input pairs closer together or push them further apart for negative input pairs. Once trained, the model creates similar embeddings for similar spikes. (b) The supervised model is based on MobileNetV2 and splits into a classification and detection branch using SSDLite. Following the classification branch, the model predicts the feature embeddings of single samples. Based on the cosine similarity loss between the predicted features and the embeddings provided by the self-supervised model, the weights of the model get adjusted. Furthermore, SSDLite predicts the score and box, compares it with ground truth (GT) by using focal loss, and adjusts the parameters accordingly. The box refers to the localisation of a spike within the sample, whereas the score prediction indicates whether a spike is present in the box or not.

to increase generalisability. Furthermore, spikes were augmented using scaling and jittering to extend the training set with similar input pairs. Each mean waveform's embedding is then used as a label for the supervised model, displayed in figure 8(b). MobileNetV2 [135] was the chosen architecture for the supervised model, as it supports edgeTPU hardware, and is well suited for applications with limited resources. The key concepts in MobileNetV2 are depth-wise separable convolutions, inverted residuals, and linear bottlenecks. The initial convolutional layer is followed by a series of bottleneck layers where the number of channels is expanded, depth-wise convolutions are carried out, and the dimensions are reduced again. The last convolutional layer is pooled, followed by a Softmax layer for classification. A single-shot detector (SSD) approach called SSDLite [135] was used to recognise spikes in the signal. SSD utilises predefined anchor boxes to predict class probabilities and box offsets at various feature map locations, refining these predictions during training with localisation and classification losses and applying non-maximum suppression (NMS) to remove overlapping boxes. SSD represents an effective alternative to the spike localisation method presented in [113], where two CNNs were utilised to detect and localise spikes in the signal, or DualSort [116], where spikes were detected, localised and classified according to their underlying neuron. However, in contrast to [116], the model in [112] does not classify spikes directly but creates a reduced feature space that is clustered subsequently. This method has the advantage of increased generalisability because the model is not restricted to certain cluster assignments or specific electrode geometry [112]. PCA was applied to reduce the feature dimensions from 32 to 10 to enhance future data adaptability and mitigate overfitting [112]. For clustering, Rokai et al carried out experiments with Isosplit5 [136], a time-efficient algorithm that does not require hyperparameters, and agglomerative clustering, which is a hierarchical method. It was demonstrated in [112] that the proposed algorithm can be used for spike detection, achieving an average accuracy of 93% on the data in [43, 62], performing better than other state-of-the-art techniques like Kilosort (77%), MountainSort4 (88%) or SpyKING CIRCUS (88%). Using two recordings of [46], the sorting accuracy of the model in [112] was 86% and 42%, whereas SpyKING CIRCUS reached 91% and 54%, respectively. As the self-supervised model processes pre-selected spike waveforms, it can be seen as input-source agnostic, which is a massive advantage in terms of flexibility [112]. One limitation of this model is the processing of overlapping spikes, as the NMS post-processing was not optimised for separating such events [112].

7. Discussion

Deep learning models have been utilised to solve specific problems in the spike sorting pipeline, which were explored in sections 3-5. Moreover, integrated systems were covered in section 6. MLPs are the simplest type of DNNs used in integrated spike sorting [116, 117]. They require few resources but have been optimised for single-channel data so far. The authors of [88, 101, 105], among others, demonstrated the advantages of convolutional layers in spike sorting. While 1D-convolutional layers were utilised to find temporal patterns in single-channel data [98, 103], 2D- convolutional layers could be used to process HD-MEA data spatiotemporally [88, 101, 105]. LSTMs have not yet been promising in detecting spikes, but could improve the temporal spike pattern recognition in some classification models [106, 113]. AEs, on the other hand, have demonstrated their effectiveness for unsupervised feature extraction [96, 103, 107], where they have been applied to reduce data dimensions to facilitate clustering. Notably, certain layer types, such as convolutional or LSTM layers, have been integrated into AEs to enhance performance [101, 103]. Furthermore, GANs have been proposed for spike classification with a semisupervised [84] and a fully unsupervised solution [94]. GANs utilise adversarial representation learning, effectively categorising spikes with minimal or no

References	Approach	Advantage	Drawback	Application
[100]	Two CNNs for channel selection and spike detection	Generalized solution to the spike sorting problem	Limited to single channel recordings	Universal channel tracking across species, brain regions and electrodes
[93]	CNN for CS detection	Accurately detects complex spikes with limited training data	Relies on several post-processing steps	Could improve neural decoding in BCIs
[96]	AE-Ensemble for feature extraction	Improved feature extraction through the use of multiple AEs	Not evaluated on overlapping spikes	Enhances spike sorting accuracy, especially with multi-dimensional inputs
[103]	Convolutional AE for feature extraction	Efficient model using low-bit weights and activations	Performance degrades in high-noise environments due to low SNR	Suitable for spike sorting in resource-constrained hardware systems
[78]	MLP for classification	Unsupervised pseudo-labelling enabled fully automatic supervised sorting	The labelling technique (PCA + k-means) limits the sorting accuracy of the MLP	Real-time spike classification
[105]	CNN for classification	Spatiotemporal sorting	Only validated on one HD-MEA dataset	On-chip sorting, enabled through a low-parameter model
[116]	MLP for integrated spike sorting	Bypasses several steps in the spike sorting pipeline	Sensitive to overlapping spikes	Real-time spike detection and sorting using a lean model architecture
[101]	VAE for integrated spike sorting	Various layer types such as convolutional and Bi-LSTM layers improve accuracy	Computationally heavy	Integrated spike detection and classification using HD-MEA data

Table 14. Key attributes of certain spike sorting algorithms: advantages, constraints and practical applications.

labelled data by disentangling label information from extracted features. This approach further reduces the need for manual intervention and is especially useful on data with limited prior information. Generally, model architectures have been optimised with regard to specific recordings. This was often done analytically, by using grid search algorithms [107], but also heuristically [78]. It must be noted that the ability of the model to generalise to new data always depends on the quality and complexity of the data on which it has been optimised, even if it has been trained on the new data.

Table 14 provides the reader with a concise descriptive table that summarises the advantages, limitations and real-world applicability of certain deep learning models discussed in this work. Table 1 lists all datasets used in the respective studies, giving an overview of which algorithms may be suitable for comparative analyses. However, comparisons between models tested on the same dataset should be made cautiously, considering that many research groups tested their models on certain parts of the datasets. Moreover, algorithms were trained using different approaches: (i) supervised; (ii) semi-supervised; (iii) unsupervised, with differently sized training sets. Depending on the technique used, this significantly impacts the results. In this survey, the results are not compared in tabular form to avoid erroneous conclusions. Instead, it discusses how the presented solutions overcome several challenges and requirements mentioned in section 1.3, points on possible future directions and presents the first hardware implementations of deep learning-based spike sorting models.

7.1. Scalability

For spike detection, models have been trained with millions of labelled waveforms and could be used universally. The authors of [85] presented a CNN for neural channel selection that can be applied to different subjects, species, brain areas and electrode types. This is beneficial for BCI applications in order to track useful channels. By using a second CNN [100], it was possible to remove background activity from the recorded channels which drastically reduced the number of waveforms for subsequent analysis. To gain insight into the computational complexity of the models in [85] and [100], the authors of this work reconstructed them using Python and TensorFlow and derived the total number of model parameters and the required floating-point operations (FLOPs). Spikedeeptector [84] and BAR [100] use 110735 and 42302 parameters and require 4157178 and

Table 15. Deep learning models trained on single channels of the synthetic data in [44] and their resource consumption.

References	Authors	Parameters	FLOPs
[78]	Park <i>et al</i>	206 595	412 163
[96]	Eom et al	2729	5353
[98]	Li et al	611 631	5788 250
[106]	Liu et al	52 414	501 060
[103]	Seong et al	27 556	430 276
[113]	Wang et al	125 957	395 968
[116]	Meyer et al	16 025	31 950

Table 16. Deep learning models trained on HD-MEA data and their resource consumption.

References	Authors	Channels	Parameters	FLOPs
[88]	Rácz et al	128	561 978	3132783
[105]	Rokai <i>et al</i>	128	5315139	275 713 102

1069 612 FLOPs, respectively. Ideally, CSs should also be detectable with BCIs, especially with implanted electrodes at the cerebellum, where CSs occur frequently. Hence, integrating CS detection into scalable multi-channel detection models, or integrated systems, is desirable for real-world applications in the future.

Certain feature extraction and classification models were also reconstructed within this survey to investigate their resource consumption. The results are displayed in table 15. Thirty-two units were assumed for the LSTM layer in [106], as this was not specified in the respective paper. Notably, all models in table 15 reached an accuracy above 95% on the utilised data. It is apparent that the ensemble learning model from Eom et al [96] requires the fewest parameters and the lowest number of FLOPs, even though Eom et al used three dense AEs. At this point, it should be noted that only the encoder modules of the AEs were reconstructed for this estimation, as the decoders are not required during inference. The resource consumption of the model in [96] is higher overall, considering the whole spike sorting pipeline, as the AE ensemble is employed only for feature extraction and additional modules for spike detection, temporal alignment, gradient determination and clustering are required. Table 15 reveals that the classification models in [78, 98, 106, 113] require significantly more parameters and FLOPs due to more complex network configurations. Notably, the size of the input vector plays a decisive role in this regard. Using the data in [44], the CNN in [98] classifies aligned spikes with a size of 64 samples, resulting in the resource consumption displayed in table 15. However, reducing the input spikes to the more 'important' samples (peak and valley), like in [116], where only twenty-five samples were used, the required memory and computation of the model in [98] shrink to 227 631 parameters (-63%) and 1552 698 FLOPs (-73%), respectively. Nevertheless,

the total resource consumptions of the above models increases by the computational effort required for spike detection and alignment. This is not the case for the model in [116], as this is an integrated system. However, since the same spike is processed several times by the same network in [116], ultimately there are more FLOPs per spike than those presented in table 15. In practice, this results in a trade-off between model robustness, gained through the described postprocessing, which also takes a small amount of computation, and resource efficiency, calibrated based on the model step size.

Scaling the discussed single-channel models up to HD-MEA recordings may significantly increase computation, and is an important subject of future research. Other classification models have already proven scalable using HD-MEA data by utilising spatiotemporal features for spike sorting. To better understand the efficiency of multi-channel models, the models from Rácz et al [88], Yi et al [105] and Rokai et al [101], were also reconstructed to estimate the required resources. The results are displayed in table 16. For both classification networks [88, 105] in table 16, it was assumed that there are twenty-seven active neurons in the recording. Table 16 demonstrates the effect of both a small number of filters and dimensionality reduction, as Yi et al [105] used only two filters and two convolutional layers, effectively shrinking dimensions. In contrast, Rácz et al [88] utilised one convolution, matched the number of filters with clusters in the data, and kept dimensions, resulting in a larger model. However, this does not imply that the model in [105] is fundamentally preferable, as the models were not tested on the same data. It may well be that the narrow architecture of the model in [105] would fail when processing the data used in [88]. Nevertheless, table 16 shows that spatiotemporal data processing can take less memory and processing power than channel-wise analyses (cf table 15).

Heavy models require external processing to ensure online sorting. Therefore, collected data must be transmitted to an external processor. In general, all described DNNs in this work can be run utilising a conventional CPU/GPU setup. Notably, the singlechannel models presented in table 15 can also process multi-channel data using modern deep learning frameworks and optimised hardware that supports parallel processing, e.g. with GPUs. CNNs benefit most from parallel processing as filters can be applied across the entire input at the same time. MLPs benefit moderately from parallel computing, while RNNs are restricted in this regard as they depend on sequential processing. Irrespective of the model type, the following must be considered: if parallel processing is applied to a supervised single-channel classification model in order to process several channels simultaneously, the model must be trained on all spikes in all channels, and the model architecture must be adjusted accordingly. ELVISort [101], is a sophisticated integrated system that requires drastically more FLOPs and parameters than the singlechannel models described above. Running ELVISort on a Windows PC using an i9-7920X with 64GB RAM and NVIDIA GeForce RTX 2080 Ti GPU, it was possible to sort spikes from 64 channels 15.71 times faster than the sampling rate [101]. Using the same setup, other state-of-the-art solutions were not that efficient: IronClust was 7.11 times faster than the sampling rate, Kilosort2 was 6.94 times faster, MountainSort4 was 2 times faster, and SpyKING CIRCUS was even slower than the sampling rate [101]. This shows that even heavy deep learning models can run faster than leading state-of-the-art solutions in spike sorting. To accelerate the performance of the model presented in [112], an edgeTPU chip was used that requires a comparatively small amount of energy (1 Watt/2 Tera Operations per second). An edgeTPU is a specialised application-specific integrated circuit (ASIC) that uses a TPU. As the postprocessing (NMS and clustering) is not supported by TPU, processing can be divided into two parts: (i) spike detection and feature extraction with TPU; (ii) NMS and sorting on CPU. As the first part requires a quantised model, the model parameters were quantised from 32-bit floating values to 8-bit integers. A quantisation-aware training method was used to minimise the performance drop during this process [112]. Data-wise, the system presented in [112] can be trained on multiple recordings simultaneously, which highly improves generalisation. To summarise, the presented setup [112] is a step towards the integration of DNN-based spike sorting for BCIs.

7.2. Hardware implementations

On-chip processing can be done using field programmable gate arrays (FPGAs) [103, 132, 137, 138] and ASICs [139–142]. Several constraints must be considered regarding implantable brain chips: the power utilisation should not exceed 10 mW, the implant size should not be higher than 1 cm², and energy dissipation should not raise the temperature of the surrounding cells by more than 1 °C [143, 144]. The BNN for single-channel classification by Valencia and Mohammad [42] was implemented in a 180-nm TSMC CMOS technology occupying an area of 0.33 mm². The ASIC only consumed 2.36 μ W with a power density of 7.15 μ W mm⁻², clearly showcasing the benefits of BNNs. It should be emphasised that BNNs mostly rely on bit-wise operations [140], making these models highly suitable for resourceconstrained environments. This significant advantage underscores the importance of directing more research and development efforts on BNNs in highbandwidth spike sorting. Other emerging hardware trends for spike sorting, for example, event-driven processors require only 2.53 μ W in 28 nm technology with an area as low as 0.018 mm² per channel [145]. Event-driven computations activate the system conditionally, thus reducing overall power consumption. NeuSort [146], a recently published neuromorphic model, uses a receptive field encoder to convert spike candidates into artificial event trains. This layer is connected to a perception layer, where firing units correspond to neural events in the data. NeuSort's efficient memorisation simplifies neural signal processing into one pass, establishing an efficient spike sorting technique [143]. The number of neuromorphic models has steadily increased in recent years and warrants a survey of its own.

Seong et al [103] proposed the first ASIC implementation of a deep learning-based feature extractor for spike sorting exhibiting a power consumption level below the permissible threshold for causing damage to brain tissue [103]. This was achieved by using 5-bit weights and 7-bit activations. Seong et al implemented 16 encoders to process 16 channels simultaneously. This design can support up to 512 channels, where each encoder processes 32 channels with time-multiplexing, all running on a 12 mm² chip. The ASIC in [103] requires an area of 38.65 mm² and consumes 8.225 mW, leading to 212.82 μ W mm⁻². In [105], a multi-channel spike classification model was proposed using weights quantisation with 8 bits for the convolutional layers and 4 bits for the dense layers, enabling a lightweight FPGA implementation. Although the accuracy dropped from 93.1% to 86.1% using quantised weights, this reduced memory requirements by 93%, going from 55 kB to 3.5 kB. Overall, the 49-channel processor required 78 mW. In [117], Zacharelos et al developed the first ASIC implementation of an integrated deep learning-based spike sorting module in a 14 nm fin field-effect transistor. They further used approximation techniques [147–149] to reduce the power consumption of their model [117]. Applied on the data in [108], Zacharelos *et al* achieved an accuracy of 93.25%, with a power and chip area of 0.434 μ W and 4787 μ m², offering a compatible, implantable design

7.3. Signal complexities

Many spike sorting algorithms struggle with varying noise levels and have particular problems with processing low SNR recordings. CAEs are known to be noise-robust, which was demonstrated in [107] with consistent results on the used datasets, but also by comparing the latent space representations of inputs with different noise levels, which showed only small deviations. Non-stationary spike waveforms, as they appear with drifting electrodes, often pose an even bigger problem for sorting algorithms, a topic that has been neglected by most research teams in deep learning-based spike sorting so far. Rácz et al [88] evaluated their supervised sorting models on relatively long recordings (45 min). They carried out experiments in which the training and test data were selected randomly and in chronological order. The chronological method (90.1% accuracy), resulted in a slightly lower performance than the random method (94.3% accuracy) [88], indicating that the model had difficulties in identifying waveforms that changed over time. Models must become more adaptive to cope with such complexities. Using resourceefficient models for spike sorting allows for iterative re-training (fine-tuning) to better adapt to timerelated signal changes. Transfer learning approaches have already been applied in applications such as EEG processing [150] to accelerate real-time adaptation and may also increase the real-world applicability of DNN-based spike sorters in the future. For supervised models, this strategy could also help to sort infrequently firing neurons. In a supervised framework, the effective classification of such events hinges on recognising a minimum quantity of a neuron's spikes and their integration into the training set. However, this may lead to strongly imbalanced classes, resulting in a biased deep learning model. Data augmentation [116, 118] is useful with sparsely firing neurons, as the number of these events can be extended to balance the data by superimposing the respective spike template with noise segments that can be captured from the recordings. However, this technique may not capture the full intra-class variance of real spikes. Unsupervised models, on the other hand, e.g. AEs that can be used to extract features, cope better with data imbalances. Nevertheless, care should be taken when choosing a clustering algorithm, as many algorithms tend to merge clusters that are easily separable but have few instances. Overlapping spikes can be seen as sparse events as well. In [91], a feature extraction method was proposed to resolve overlapping spikes in the feature space by using a custom cost function. One limitation of this method is that the exact spike times of overlapping instances cannot

be determined, however, depending on the specific application at hand, this issue may not necessarily pose a concern [91]. In contrast to the method in [91], the authors of [106] and [115] did not try to separate colliding waveforms but taught their models how overlaps of specific waveforms appear in the signal. Even if this method yields high results, its usability is limited to channels with few active neurons. For channels containing spikes from up to 20 actively firing neurons, as in the synthetic data in [2], the training set would be impractically huge, as this model would require 20 classes to indicate single spike firings and another 190 classes to indicate all possible collisions, considering that only two spikes can fire simultaneously. Scaling this approach up to HD-MEA recordings is an even bigger challenge. Nevertheless, the problem of overlapping spikes can be alleviated to some extent by analysing HD-MEA recordings spatiotemporally, as spatial data provides additional information that can be used to assign spikes to their neurons.

7.4. Limited prior information

Generally, unsupervised methods are preferred over supervised methods as there is limited prior information about the data distribution in neural recordings. However, with an increasing number of active neurons in the signals, the unsupervised methods in [96, 103, 111] struggled to determine the number of firing neurons in the recording. In the context of spike classification, supervised deep learning models require specific information about the number of active neurons in the recording and need labelled examples of spikes of each neuron. The standard approach to tackle this problem is illustrated in figure 5, where an unsupervised method is applied offline to obtain pseudo-labels that can be utilised to train a supervised model. Using standard methods such as PCA and k-means, as done in [78], often results in an accuracy drop of the supervised model when faced with highly correlated spikes. Moreover, the k-means algorithm requires the number of clusters k, and therefore still relies on additional techniques, such as gap statistics, to run automatically. Supervised multi-channel models [88, 105], on the other side, were utilising more sophisticated software to obtain pseudo-labels. By using Kilosort, 23-46 neurons were isolated in the HD-MEA recordings [89] used in [88] and [101]. However, recently, the spikes of this data were re-sorted utilising Kilosort2, which isolated 73–183 units for the same recordings. Hence, the models in [88] and [101] were presumably trained and evaluated with merged clusters (similar waveforms from different neurons), underscoring the problem of obtaining GT in extracellular recordings. Combining a supervised and self-supervised training strategy [112], proved to be a practical method to overcome the problem of limited prior information to **IOP** Publishing

some extent. This strategy enhances the model's generalisability, allowing a more accurate processing of new recordings. In contrast, using a solely supervised strategy usually fails because experimental recordings, after sorting and manual curation, provide labels that typically include information about the specific channel position, making it challenging to address the entirety of the spike sorting pipeline with an onlysupervised training strategy.

8. Conclusion

In recent years, different types of DNNs have been developed to address various spike sorting challenges. Currently, no single solution solves all the difficulties associated with spike sorting, but the methods discussed in this work build a solid base for future model development. By categorising all investigated studies into spike detection, feature extraction, spike classification and integrated systems, this survey provides a comprehensive overview of contemporary DNNbased spike sorting models. As evident in table 1, CNNs have been utilised extensively in spike detection. Training the model in [85] on a large dataset of a single subject enabled neural channel tracking across subjects, in different brain areas, species, and even with varying electrode types, showcasing that a supervised model trained on a large dataset can be used universally. Moreover, multi-channel processing CNNs for detecting spike waveforms already exist and were embedded into established spike sorting pipelines [107, 118]. However, certain cell types in specific brain regions may generate CSs that can be missed if the model has not been explicitly trained on such events [93]. Inclusive models that can manage all types of spikes reliably do not exist so far. Hence, multiple computationally heavy models would be required to detect spikes of all kind. AEs were mainly utilised to extract features, with ensemble models [96] emerging as a valuable technique to identify different waveform characteristics [44]. Furthermore, convolutional layers within the AEs [101] and a contractive loss term [107] have proven useful to capture spatiotemporal patterns and gain noise robustness, respectively. An inherently compelling rationale supporting the utilisation of AEs is their potential for unsupervised learning, offering a significant edge over supervised systems in situations with limited prior information. This capability could be used for dimensionality reduction [96, 101, 103], which often turned clustering into a trivial problem, and for spike classification, by using adversarial representation learning [94]. Additionally, offline clustering with unsupervised sorters has been applied to generate pseudo-labels for supervised classification models [78, 98, 106]. Although many of these models could sort spikes at a high level, most are computationally expensive and have not yet been

applied to HD-MEA data. However, the proposed multi-channel classification model [105] consumed even fewer resources than the single-channel models mentioned above, enabling efficient sorting with an FPGA [105]. In this regard, model quantisation plays a pivotal role in reducing the required memory and computation. While most models to date have been developed for specific tasks in spike sorting, integrated systems have already shown promising results by addressing the spike sorting problem with end-to-end models [101, 116]. In many studies analysed, deep learning-based spike sorting models outperformed state-of-the-art solutions like SPC [44], Kilosort [21] and MountainSort [59]. Remarkably, even complex DNNs with several layer types and many hidden layers [100] run faster in a CPU/GPU environment than the state-of-the-art spike sorting algorithms mentioned above. Integrating these methods into more adaptive and scalable systems that can handle a diverse range of signal complexities, ideally by using implantable hardware and performing realtime processing at scale will be crucial in meeting the computational demands of future neuroscience research and next-generation neurotechnology.

Data availability statement

The data that support the findings of this study are available upon reasonable request from the authors.

ORCID iDs

Luca M Meyer lo https://orcid.org/0000-0002-5332-2463

Majid Zamani lhttps://orcid.org/0009-0007-0844-473X

János Rokai l https://orcid.org/0000-0001-8606-298X

Andreas Demosthenous in https://orcid.org/0000-0003-0623-963X

References

- Hodgkin A L and Huxley A F 1952 A quantitative description of membrane current and its application to conduction and excitation in nerve J. Physiol. 117 500–44
- [2] Pedreira C, Martinez J, Ison M J and Quian Quiroga R 2012 How many neurons can we see with current spike sorting algorithms? J. Neurosci. Methods 211 58–65
- [3] Henze D A, Borhegyi Z, Csicsvari J, Mamiya A, Harris K D and Buzsáki G 2000 Intracellular features predicted by extracellular recordings in the hippocampus in vivo J. Neurophysiol. 84 390–400
- [4] Buzsáki G 2004 Large-scale recording of neuronal ensembles Nat. Neurosci. 7 446–51
- [5] Gibson S, Judy J W and Markovic D 2012 Spike sorting: the first step in decoding the brain *IEEE Signal Process. Mag.* 29 124–43
- [6] Gold C, Henze D A, Koch C and Buzsáki G 2006 On the origin of the extracellular action potential waveform: a modeling study J. Neurophysiol. 95 3113–28

- [7] Jun J J *et al* 2017 Fully integrated silicon probes for high-density recording of neural activity *Nature* 551 232–6
- [8] Nicolelis M A L 2001 Actions from thoughts Nature 409 403–7
- [9] Hochberg L R, Serruya M D, Friehs G M, Mukand J A, Saleh M, Caplan A H, Branner A, Chen D, Penn R D and Donoghue J P 2006 Neuronal ensemble control of prosthetic devices by a human with tetraplegia *Nature* 442 164–71
- [10] Berger T *et al* 2005 Restoring lost cognitive function *IEEE* Eng. Med. Biol. Mag. 24 30–44
- [11] Shenoy K V, Meeker D, Cao S, Kureshi S A, Pesaran B, Buneo C A, Batista A P, Mitra P P, Burdick J W and Andersen R A 2003 Neural prosthetic control signals from plan activity *Neuro Rep.* 14 591–6
- [12] Ebrahimighahnavieh M A, Luo S and Chiong R 2020 Deep learning to detect Alzheimer's disease from neuroimaging: a systematic literature review *Comput. Methods Programs Biomed.* 187 105242
- [13] Karthik R, Menaka R, Johnson A and Anand S 2020 Neuroimaging and deep learning for brain stroke detection—A review of recent advancements and future prospects *Comput. Methods Programs Biomed.* 197 105728
- [14] Khodatars M *et al* 2021 Deep learning for neuroimaging-based diagnosis and rehabilitation of autism spectrum disorder: a review *Comput. Biol. Med.* 139 104949
- [15] Vieira S, Pinaya W H L and Mechelli A 2017 Using deep learning to investigate the neuroimaging correlates of psychiatric and neurological disorders: methods and applications *Neurosci. Biobehav. Rev.* 74 58–75
- [16] Işın A, Direkoğlu C and Şah M 2016 Review of MRI-based brain tumor image segmentation using deep learning methods Proc. Comput. Sci. 102 317–24
- [17] Mishro P K, Agrawal S, Panda R and Abraham A 2022 A survey on state-of-the-art denoising techniques for brain magnetic resonance images *IEEE Rev. Biomed. Eng.* 15 184–99
- [18] Tuladhar A, Moore J A, Ismail Z and Forkert N D 2021 Modeling neurodegeneration in silico with deep learning *Front. Neuroinform.* 15 748370
- [19] Livezey J A and Glaser J I 2021 Deep learning approaches for neural decoding across architectures and recording modalities *Brief. Bioinform.* 22 1577–91
- [20] Craik A, He Y and Contreras-Vidal J L 2019 Deep learning for electroencephalogram (EEG) classification tasks: a review J. Neural Eng. 16 031001
- [21] Pachitariu M, Steinmetz N, Kadir S, Carandini M and Kenneth D H 2016 Kilosort: realtime spike-sorting for extracellular electrophysiology with hundreds of channels *bioxRiv Preprint* (https://doi.org/10.1101/061481) (posted online 30 June 2016, accessed 2 February 2023)
- [22] Lewicki M S 1998 A review of methods for spike sorting: the detection and classification of neural action potentials *Netw., Comput. Neural Syst.* 9 R53–R78
- [23] Einevoll G T, Franke F, Hagen E, Pouzat C and Harris K D 2012 Towards reliable spike-train recordings from thousands of neurons with multielectrodes *Curr. Opin. Neurobiol.* 22 11–17
- [24] Rey H G, Pedreira C and Quian Quiroga R 2015 Past, present and future of spike sorting techniques *Brain Res. Bull.* 119 106–17
- [25] Lefebvre B, Yger P and Marre O 2016 Recent progress in multi-electrode spike sorting methods J. Physiol. Paris 110 327–35
- [26] Zhang T, Azghadi M R, Lammie C, Amirsoleimani A and Genov R 2023 Spike sorting algorithms and their efficient hardware implementation: a comprehensive survey J. *Neural Eng.* 20 021001
- [27] Buccino A P, Garcia S and Yger P 2022 Spike sorting: new trends and challenges of the era of high-density probes *Prog. Biomed. Eng.* 4 022005

- [28] Bod R B, Rokai J, Meszéna D, Fiáth R, Ulbert I and Márton G 2022 From end to end: gaining, sorting, and employing high-density neural single unit recordings *Front. Neuroinform.* 16 851024
- [29] Hussein H A, Zeebaree S R M, Sadeeq M A M, Shukur H M, Alkhayyat A and Sharif K H 2021 An investigation on neural spike sorting algorithms 2021 Int. Conf. on Communication and Information Technology (ICICT) (IEEE) pp 202–7
- [30] Kim K H and Kim S J 2000 Neural spike sorting under nearly 0-dB signal-to-noise ratio using nonlinear energy operator and artificial neural-network classifier *IEEE Trans. Biomed. Eng.* 47 1406–11
- [31] Nenadic Z and Burdick J W 2005 Spike detection using the continuous wavelet transform *IEEE Trans. Biomed. Eng.* 52 74–87
- [32] Pearson K 1901 On lines and planes of closest fit to systems of points in space London, Edinburgh Dublin Phil. Mag. J. Sci. 2 559–72
- [33] Comon P 1994 Independent component analysis, A new concept? Signal Process. 36 287–314
- [34] Dimitriadis G, Neto J P and Kampff A R 2018 t-SNE visualization of large-scale neural recordings *Neural Comput.* 30 1750–74
- [35] Letelier J C and Weber P P 2000 Spike sorting based on discrete wavelet transform coefficients J. Neurosci. Methods 101 93–106
- [36] Lloyd S 1982 Least squares quantization in PCM IEEE Trans. Inf. Theory 28 129–37
- [37] Ester M, Kriegel HP, Sander J and Xu X 1996 A density-based algorithm for discovering clusters in large spatial databases with noise *Proc. Second Int. Conf. on Knowledge Discovery and Data Mining, in KDD'96* (AAAI Press) pp 226–31
- [38] Geng X, Hu G and Tian X 2010 Neural spike sorting using mathematical morphology, multiwavelets transform and hierarchical clustering *Neurocomputing* 73 707–15
- [39] Bar-hillel A, Spiro A and Stark E 2004 Spike sorting: bayesian clustering of non-stationary data Advances in Neural Information Processing Systems (MIT Press) (https:// doi.org/10.1016/j.jneumeth.2006.04.023)
- [40] Vogelstein R J, Murari K, Thakur P H, Diehl C, Chakrabartty S and Cauwenberghs G 2004 Spike sorting with support vector machines *The 26th Annual Int. Conf. IEEE Engineering in Medicine and Biology Society* (IEEE) pp 546–9
- [41] Issar D, Williamson R C, Khanna S B and Smith M A 2020 A neural network for online spike classification that improves decoding accuracy J. Neurophysiol. 123 1472–85
- [42] Valencia D and Alimohammad A 2021 Neural spike sorting using binarized neural networks *IEEE Trans. Neural Syst. Rehabil. Eng.* 29 206–14
- [43] Allen B D et al 2018 Automated in vivo patch-clamp evaluation of extracellular multielectrode array spike recording capability J. Neurophysiol. 120 2182–200
- [44] Quiroga R Q, Nadasdy Z and Ben-Shaul Y 2004 Unsupervised spike detection and sorting with wavelets and superparamagnetic clustering *Neural Comput.* 16 1661–87
- [45] Teeters J L, Harris K D, Millman K J, Olshausen B A and Sommer F T 2008 Data sharing for computational neuroscience *Neuroinform* 6 47–55
- [46] Magland J, Jun J J, Lovero E, Morley A J, Hurwitz C L, Buccino A P, Garcia S and Barnett A H 2020 SpikeForest, reproducible web-facing ground-truth validation of automated neural spike sorters *eLife* 9 e55167
- [47] Database from the Centre for Systems Neuroscience University of Leicester (available at: https://le.ac.uk/csn/ data) (Accessed 1 May 2024)
- [48] Datasets of extracellular recordings *UlbertLab* (available at: www.ulbertlab.com/datasets) (Accessed 1 May 2024)
- [49] Fee M S, Mitra P P and Kleinfeld D 1996 Variability of extracellular spike waveforms of cortical neurons J. *Neurophysiol.* 76 3823–33

- [50] Adamos D A, Laskaris N A, Kosmidis E K and Theophilidis G 2010 NASS: an empirical approach to spike sorting with overlap resolution based on a hybrid noise-assisted methodology *J. Neurosci. Methods* 190 129–42
- [51] Choi J H, Kim D and Kim T 2005 A new overlapping resolution method for multi-channel spike sorting *Conf. Proc. 2nd Int. IEEE EMBS Conf. on Neural Engineering*, 2005 pp 683–6
- [52] Sauer I, Doerr C and Schanze T 2015 Spike sorting: the overlapping spikes challenge *Curr. Direct Biomed. Eng.* 1 42–45
- [53] Rutishauser U, Schuman E M and Mamelak A N 2006 Online detection and sorting of extracellularly recorded action potentials in human medial temporal lobe recordings, *in vivo J. Neurosci. Methods* 154 204–24
- [54] Mohammadi Z, Denman D J, Klug A and Lei T C 2024 A fully automatic multichannel neural spike sorting algorithm with spike reduction and positional feature J. *Neural Eng.* 21 046039
- [55] Shaeri M and Sodagar A M 2020 A framework for on-implant spike sorting based on salient feature selection *Nat. Commun.* 11 3278
- [56] Chen Y, Tacca B, Chen Y, Biswas D, Gielen G, Catthoor F, Verhelst M and Mora Lopez C 2023 An online-spikesorting IC using unsupervised geometry-aware OSort clustering for efficient embedded neural-signal processing *EEE J. Solid-State Circuits* 58 2990–3002
- [57] Carlson D and Carin L 2019 Continuing progress of spike sorting in the era of big data *Curr. Opin. Neurobiol.* 55 90–96
- [58] Pachitariu M, Sridhar S, Pennington J and Stringer C 2024 Spike sorting with Kilosort4 Nat. Methods 21 914–21
- [59] Chung J E, Magland J F, Barnett A H, Tolosa V M, Tooker A C, Lee K Y, Shah K G, Felix S H, Frank L M and Greengard L F 2017 A fully automated approach to spike sorting *Neuron* 95 1381–94.e6
- [60] Magland J MountainSort4 spike sorting algorithm (available at: https://github.com/magland/mountainsort4) (Accessed 14 May 2024)
- [61] Rossant C et al 2016 Spike sorting for large, dense electrode arrays Nat. Neurosci. 19 634–41
- [62] Yger P et al 2018 A spike sorting toolbox for up to thousands of electrodes validated with ground truth recordings in vitro and in vivo elife 7 e34518
- [63] Jun J J, Mitelut C, Lai C, Gratiy S L, Anastassiou C A and Harris T D 2017 Real-time spike sorting platform for high-density extracellular probes with ground-truth validation and drift correction *bioxRiv Preprint* (https:// doi.org/10.1101/101030) (posted online 30 January 2017, accessed 5 January 2024)
- [64] Pachitariu M, Sridhar S and Stringer C 2023 Solving the spike sorting problem with Kilosort *bioRxiv Preprint* (https://doi.org/10.1101/2023.01.07.523036) (posted online 7 January 2023, accessed 23 July 2023)
- [65] Bower J M, Wong Y F and Banik J 1988 Neural networks for template matching: application to real-time classification of the action potentials of real neurons
- [66] Jansen R F 1990 The reconstruction of individual spike trains from extracellular multineuron recordings using a neural network emulation program J. Neurosci. Methods 35 203–13
- [67] Yamada S, Kage H, Nakashima M, Shiono S and Maeda M 1992 Data processing for multi-channel optical recording: action potential detection by neural network *J. Neurosci. Methods* 43 23–33
- [68] Mirfakhraei K and Horch K 1994 Classification of action potentials in multi-unit intrafascicular recordings using neural network pattern-recognition techniques *IEEE Trans. Biomed. Eng.* 41 89–91
- [69] Hermle T, Schwarz C and Bogdan M 2004 Employing ICA and SOM for spike sorting of multielectrode recordings from CNS J. Physiol. Paris 98 349–56

- [70] Hermle T, Bogdan M, Schwarz C and Rosenstiel W 2005 ANN-based system for sorting spike waveforms employing refractory periods Artificial Neural Networks: Biological Inspirations—ICANN 2005 (Lecture Notes in Computer Science) ed W Duch, J Kacprzyk, E Oja and S Zadrożny (Springer) pp 121–6
- [71] Horton P M, Nicol A U, Kendrick K M and Feng J F 2007 Spike sorting based upon machine learning algorithms (SOMA) J. Neurosci. Methods 160 52–68
- [72] Werner T, Vianello E, Bichler O, Garbin D, Cattaert D, Yvert B, De Salvo B and Perniola L 2016 Spiking neural networks based on OxRAM synapses for realtime unsupervised spike sorting *Front. Neurosci.* 10 474
- [73] Bernert M and Yvert B 2017 Fully unsupervised online spike sorting based on an artificial spiking neural network *Neuroscience* (https://doi.org/10.1101/236224) Preprint
- [74] Mukhopadhyay A K, Chakrabarti I, Basu A and Sharad M 2018 Power efficient spiking neural network classifier based on memristive crossbar network for spike sorting application (arXiv:1802.09047)
- [75] Bernert M and Yvert B 2019 An attention-based spiking neural network for unsupervised spike-sorting *Int. J. Neural Syst.* 29 1850059
- [76] Wang J 2022 A review of spiking neural networks SHS Web Conf. 144 03004
- [77] Cybenko G 1989 Approximation by superpositions of a sigmoidal function Math. Control Signals Syst. 2 303–14
- [78] Park I Y, Eom J, Jang H, Kim S, Park S, Huh Y and Hwang D 2019 Deep learning-based template matching spike classification for extracellular recordings *Appl. Sci.* **10** 301
- [79] Lee J H et al 2017 YASS: yet another spike sorter Advances in Neural Information Processing Systems (Curran Associates, Inc) (https://doi.org/10.1101/151928)
- [80] Hagen E, Ness T V, Khosrowshahi A, Sørensen C, Fyhn M, Hafting T, Franke F and Einevoll G T 2015 ViSAPy: a Python tool for biophysics-based generation of virtual spiking activity for evaluation of spike-sorting algorithms J. Neurosci. Methods 245 182–204
- [81] Litke A M et al 2004 What does the eye tell the brain?: development of a system for the large-scale recording of retinal output activity *IEEE Trans. Nucl. Sci.* 51 1434–40
- [82] Yang K, Wu H and Zeng Y 2017 A simple deep learning method for neuronal spike sorting *J. Phys.: Conf. Ser.* 910 012062
- [83] Martinez J, Pedreira C, Ison M J and Quian Quiroga R 2009 Realistic simulation of extracellular recordings J. Neurosci. Methods 184 285–93
- [84] Wu T, Ratkai A, Schlett K, Grand L and Yang Z 2019 Learning to sort: few-shot spike sorting with adversarial representation learning 2019 41st Annual Int. Conf. IEEE Engineering in Medicine and Biology Society (EMBC) (IEEE) pp 713–6
- [85] Saif-ur-Rehman M *et al* 2019 SpikeDeeptector: a deep-learning based method for detection of neural spiking activity *J. Neural Eng.* 16 056003
- [86] Shi Y, Apker G and Buneo C A 2013 Multimodal representation of limb endpoint position in the posterior parietal cortex *J. Neurophysiol.* 109 2097–107
- [87] Lawlor P N, Perich M G, Miller L E and Kording K P 2018 Linear-nonlinear-time-warp-poisson models of neural activity J. Comput. Neurosci. 45 173–91
- [88] Rácz M, Liber C, Németh E, Fiáth R, Rokai J, Harmati I, Ulbert I and Márton G 2020 Spike detection and sorting with deep learning *J. Neural Eng.* 17 016038
- [89] Fiáth R, Márton A L, Mátyás F, Pinke D, Márton G, Tóth K and Ulbert I 2019 Slow insertion of silicon probes improves the quality of acute neuronal recordings *Sci. Rep.* 9 111
- [90] Huh Y and Cho J 2016 Differential responses of thalamic reticular neurons to nociception in freely behaving mice *Frontiers Behav. Neurosci.* 10 223
- [91] Wouters J, Kloosterman F and Bertrand A 2020 A neural network-based spike sorting feature map that resolves spike

overlap in the feature space ICASSP 2020–2020 IEEE Int. Conf. on Acoustics, Speech and Signal Processing (ICASSP) pp 1175–9

- [92] Steinmetz N, Carandini M and Harris K D 'Single Phase3' and 'Dual Phase3' Neuropixels Datasets (available at: https://figshare.com/articles/dataset/_Single_ Phase3_Neuropixels_Dataset/7666892) (Accessed 6 February 2024)
- [93] Markanday A, Bellet J, Bellet M E, Inoue J, Hafed Z M and Thier P 2020 Using deep neural networks to detect complex spikes of cerebellar Purkinje cells *J. Neurophysiol.* 123 2217–34
- [94] Ciecierski K 2020 Neural spike sorting using unsupervised adversarial learning Presented at the 25th Int. Symp. on Methodologies for Intelligent Systems (https://doi.org/ 10.1007/978-3-030-59491-6_18)
- [95] Israel Z and Burchiel J 2004 Microelectrode Recording in Movement Disorder Surgery 2004th (Thieme Verlag)
- [96] Eom J, Park I Y, Kim S, Jang H, Park S, Huh Y and Hwang D 2021 Deep-learned spike representations and sorting via an ensemble of auto-encoders *Neural Netw.* 134 131–42
- [97] Huh Y and Cho J 2013 Discrete pattern of burst stimulation in the ventrobasal thalamus for anti-nociception *PLoS One* 8 e67655
- [98] Li Z, Wang Y, Zhang N and Li X 2020 An accurate and robust method for spike sorting based on convolutional neural networks *Brain Sci.* 11 835
- [99] Chu C C J, Chien P F and Hung C P 2014 Tuning dissimilarity explains short distance decline of spontaneous spike correlation in macaque V1 Vis. Res. 96 113–32
- [100] Saif-ur-Rehman M *et al* 2021 SpikeDeep-classifier: a deep-learning based fully automatic offline spike sorting algorithm J. Neural Eng. 18 016009
- [101] Rokai J, Rácz M, Fiáth R, Ulbert I and Márton G 2021 ELVISort: encoding latent variables for instant sorting, an artificial intelligence-based end-to-end solution *J. Neural Eng.* 18 046033
- [102] Neto J P et al 2016 Validating silicon polytrodes with paired juxtacellular recordings: method and dataset J. Neurophysiol. 116 892–903
- [103] Seong C, Lee W and Jeon D 2021 A multi-channel spike sorting processor with accurate clustering algorithm using convolutional autoencoder *IEEE Trans. Biomed. Circuits Syst.* 15 1441–53
- [104] Yang Y, Boling S and Mason A J 2017 A hardware-efficient scalable spike sorting neural signal processor module for implantable high-channel-count brain machine interfaces *IEEE Trans. Biomed. Circuits Syst.* **11** 743–54
- [105] Yi J, Xu J, Chen E, Chamanzar M and Chen V 2022 Multichannel many-class real-time neural spike sorting with convolutional neural networks *IEEE Open J. Circuits Syst.* 3 168–79
- [106] Liu M, Feng J, Wang Y and Li Z 2022 Classification of overlapping spikes using convolutional neural networks and long short term memory *Comput. Biol. Med.* 148 105888
- [107] Radmanesh M, Rezaei A A, Jalili M, Hashemi A and Goudarzi M M 2022 Online spike sorting via deep contractive autoencoder *Neural Netw.* 155 39–49
- [108] Camuñas-Mesa L A and Quiroga R Q 2013 A detailed and fast model of extracellular recordings *Neural Comput.* 25 1191–212
- [109] Okreghe C O, Zamani M and Demosthenous A 2023 A deep neural network-based spike sorting with improved channel selection and artefact removal *IEEE Access* 11 15131–43
- [110] Buneo C A, Shi Y, Apker G and VanGilder P 2016 Multimodal Spike Data Recorded from Posterior Parietal Cortex of Non-Human Primates Performing a Reaction-Time Task Involving Combined Eye and Arm Movements While in a Virtual Reality Environment (CRCNS.org) (https://doi. org/10.6080/K0CZ353K)

- [111] Ardelean ER, Coporîie A, Ichim AM, Dînşoreanu M and Mureşan R C 2023 A study of autoencoders as a feature extraction technique for spike sorting *PLoS One* 18 e0282810
- [112] Rokai J, Ulbert I and Márton G 2023 Edge computing on TPU for brain implant signal analysis *Neural Netw.* 162 212–24
- [113] Wang M, Zhang L, Yu H, Chen S, Zhang X, Zhang Y and Gao D 2023 A deep learning network based on CNN and sliding window LSTM for spike sorting *Comput. Biol. Med.* 159 106879
- [114] Wild J, Prekopcsák Z, Sieger T, Novak D and Jech R 2012 Performance comparison of extracellular spike sorting algorithms for single-channel recordings J. Neurosci. Methods 203 369–76
- [115] Zhang L, Gao D and Wang M 2023 Sorting overlapping spikes based on log-mel spectrogram and convolutional neural networks 2023 6th Int. Conf. on Artificial Intelligence and Big Data (ICAIBD) (IEEE) pp 482–5
- [116] Meyer L M, Samann F and Schanze T 2023 DualSort: online spike sorting with a running neural network J. Neural Eng. 20 056031
- [117] Zacharelos E, Scognamillo C, Napoli E and Gragnaniello D 2023 On-chip spike detection and classification using neural networks and approximate computing 2023 IEEE Biomedical Circuits and Systems Conf. (BioCAS) pp 1–5
- [118] Lee J et al 2020 YASS: yet Another Spike Sorter applied to large-scale multi-electrode array recordings in primate retina Neuroscience (https://doi.org/10.1101/2020. 03.18.997924)
- [119] Ioffe S and Szegedy C 2015 Batch normalization: accelerating deep network training by reducing internal covariate shift (arXiv:1502.03167)
- [120] Srivastava N, Hinton G, Krizhevsky A, Sutskever I and Salakhutdinov R 2014 Dropout: a simple way to prevent neural networks from overfitting *J. Mach. Learn. Res.* 15 1929–58
- [121] Leznik E and Llinás R 2005 Role of gap junctions in synchronized neuronal oscillations in the inferior olive J. Neurophysiol. 94 2447–56
- [122] Albus J S 1971 A theory of cerebellar function *Math. Biosci.* 10 25–61
- [123] Muller S Z, Pi J S, Hage P, Fakharian M A, Sedaghat-Nejad E and Shadmehr R 2023 Complex spikes perturb movements, revealing the sensorimotor map of Purkinje cells *bioRxiv Preprint* (https://doi.org/10.1101/ 2023.04.16.537034) (posted online 16 April 2023, accessed 9 October 2023)
- [124] Ronneberger O, Fischer P and Brox T 2015 U-Net: convolutional networks for biomedical image segmentation Medical Image Computing and Computer-Assisted Intervention—MICCAI 2015 (Lecture Notes in Computer Science) ed N Navab, J Hornegger, W M Wells and A F Frangi (Springer International Publishing) pp 234–41
- [125] McInnes L, Healy J and Melville J 2020 UMAP: uniform manifold approximation and projection for dimension reduction (arXiv:1802.03426)
- [126] Catz N, Dicke P W and Thier P 2005 Cerebellar complex spike firing is suitable to induce as well as to stabilize motor learning *Curr. Biol.* 15 2179–89
- [127] Kechris C, Delitzas A, Matsoukas V and Petrantonakis P C 2021 Removing noise from extracellular neural recordings using fully convolutional denoising autoencoders Annu Int. Conf. IEEE Engineering in Medicine and Biology Society vol 2021 pp 890–3
- [128] Wu T, Zhao W, Keefer E and Yang Z 2018 Deep compressive autoencoder for action potential compression in large-scale neural recording *J. Neural Eng.* 15 066019
- [129] Walczak S and Cerpa N 1999 Heuristic principles for the design of artificial neural networks *Inf. Softw. Technol.* 41 107–17

- [130] Nadian M H, Karimimehr S, Doostmohammadi J, Ghazizadeh A and Lashgari R 2018 A fully automated spike sorting algorithm using t-distributed neighbor embedding and density based clustering *Neuroscience* (https://doi.org/ 10.1101/418913) Preprint
- [131] Manton J, Applebaum D, Ikeda S and Le Bihan N 2013 Introduction to the issue on differential geometry in signal processing *IEEE J. Sel. Top. Signal Process.* 7 573–5
- [132] Li P, Liu M, Zhang X and Chen H 2014 Efficient online feature extraction algorithm for spike sorting in a multichannel FPGA-based neural recording system 2014 IEEE Biomedical Circuits and Systems Conf. (BioCAS) Proc. pp 1–4
- [133] Huang L, Ling B WK, Cai R, Zeng Y, He J and Chen Y 2019 WMsorting: wavelet packets' decomposition and mutual information-based spike sorting method *IEEE Trans. Nanobiosci.* 18 283–95
- [134] Dwibedi D, Aytar Y, Tompson J, Sermanet P and Zisserman A 2021 With a little help from my friends: nearest-neighbor contrastive learning of visual representations 2021 IEEE/CVF Int. Conf. on Computer Vision (ICCV) (IEEE) pp 9568–77
- [135] Sandler M, Howard A, Zhu M, Zhmoginov A and Chen LC 2019 MobileNetV2: inverted residuals and linear bottlenecks (arXiv:1801.04381)
- [136] Magland J F and Barnett A H 2016 Unimodal clustering using isotonic regression: ISO-SPLIT (arXiv:1508.04841)
- [137] Park J, Kim G and Jung SD 2017 A 128-channel FPGA-based real-time spike-sorting bidirectional closed-loop neural interface system *IEEE Trans. Neural Syst. Rehabil. Eng.* 25 2227–38
- [138] Schäffer L, Nagy Z, Kincses Z and Fiáth R 2017 "FPGA-based neural probe positioning to improve spike sorting with OSort algorithm *IEEE Int. Symp. on Circuits* and Systems (https://doi.org/10.1109/ISCAS.2017.8050608)
- [139] Valencia D and Alimohammad A 2019 An efficient hardware architecture for template matching-based spike sorting *IEEE Trans. Biomed. Circuits Syst.* 13 481–92
- [140] Zamani M, Sokolić J, Jiang D, Renna F, Rodrigues M R D and Demosthenous A 2020 Accurate, very low

computational complexity spike sorting using unsupervised matched subspace learning *IEEE Trans. Biomed. Circuits Syst.* **14** 221–31

- [141] Zamani M, Jiang D and Demosthenous A 2018 An adaptive neural spike processor with embedded active learning for improved unsupervised sorting accuracy *IEEE Trans. Biomed. Circuits Syst.* **12** 665–76
- [142] Valencia D and Alimohammad A 2019 A real-time spike sorting system using parallel OSort clustering *IEEE Trans. Biomed. Circuits Syst.* 13 1700–13
- [143] Sarkar S 2023 Advanced spike sorting approaches in implantable VLSI wireless brain computer interfaces: a survey (arXiv:2309.00913)
- [144] Esfahani E T and Sundararajan V 2011 Using brain–computer interfaces to detect human satisfaction in human–robot interaction *Int. J. Hum. Robot.* 08 87–101
- [145] Jiang H et al 2023 A 2.53 microWatt/channel event- driven neural spike sorting processor with sparsity- aware computing-in-memory macros 2023 IEEE Int. Symp. on Circuits and Systems (ISCAS) pp 1–5
- [146] Yu H, Qi Y and Pan G 2023 NeuSort: an automatic adaptive spike sorting approach with neuromorphic models J. *Neural Eng.* 20 056006
- [147] Strollo A G M, Napoli E, De Caro D, Petra N, Saggese G and Di Meo G 2022 Approximate multipliers using static segmentation: error analysis and improvements *IEEE Trans. Circuits Syst.* I 69 2449–62
- [148] Zacharelos E, Nunziata I, Saggese G, Strollo A G M and Napoli E 2022 Approximate recursive multipliers using low power building blocks *IEEE Trans. Emerg. Top. Comput.* 10 1315–30
- [149] Esposito D, Strollo A G M, Napoli E, De Caro D and Petra N 2018 Approximate multipliers based on new approximate compressors *IEEE Trans. Circuits Syst.* I 65 4169–82
- [150] Raghu S, Sriraam N, Temel Y, Rao S V and Kubben P L 2020 EEG based multi-class seizure type classification using convolutional neural network and transfer learning *Neural Netw.* 124 202–12